



Article A Conceptual Blockchain Enhanced Information Model of Product Service Systems Framework for Sustainable Furniture

Jing Liu¹, Zhen Liu¹, Qiong Yang^{2,*}, Mohamed Osmani³ and Peter Demian³

- ¹ School of Design, South China University of Technology, Guangzhou 510006, China
- ² Academic Affairs Office, South China University of Technology, Guangzhou 510641, China
- ³ School of Architecture, Building and Civil Engineering, Loughborough University, Loughborough LE11 3TU, UK
- * Correspondence: jwqyang@scut.edu.cn

Abstract: The sustainable development of the furniture industry is experiencing the challenges of energy consumption and waste disposal. Product-service systems (PSSs) have the potential to promote sustainable development and the opportunity to transition to a circular economy (CE). PSSs can bring a series of benefits to sustainable furniture, but there are still some problems to be considered, such as the safe storage and transmission of information and data and the protection of stakeholders' rights and interests. With digitization becoming a major trend, emerging digital technologies such as blockchain (BC) are proving to have the potential to solve related problems. Therefore, this paper aims to integrate the potential roles of BC and PSSs in the lifecycle of sustainable furniture. This paper adopts a mixed quantitative and qualitative research method. Firstly, the potential relationship among furniture, PSSs, and BC was quantitatively analyzed by VOSviewer. Secondly, this paper qualitatively analyzes the lifecycle stages of sustainable furniture, the advantages of PSSs to promote sustainable furniture, and the potential of BC to enhance the PSSs information model (IM) to further promote sustainable furniture to address related challenges. Subsequently, the conceptual BC-enhanced PSSs IM (BC-PSSs) framework was constructed, which contains the high-level and the low-level of structure and process, and then reviewed and refined through pre-interview questionnaires and follow-up interviews by industry experts and scholars. In addition, discussing the contribution of the conceptual BC-PSSs framework in sustainable furniture, and the potential of BC-PSSs in quantifying design value, encouraging designers to contribute value, and exploring the potential role of BC-PSSs in supporting sustainable consumer behavior. It is the first attempt to construct a conceptual BC-enhanced PSSs IM framework for sustainable furniture from the perspective of lifecycle stages, which can serve as a reference for researchers and policymakers in relevant directions to support sustainable development, in particular contributing to the achievement of SDGs 11 (Sustainable Cities and Communities) and SDGs 12 (Responsible Consumption and Production).

Keywords: product–service systems (PSSs); information model; blockchain; sustainability; furniture; lifecycle management; circular economy (CE); consumer behavior; design

1. Introduction

With the devastating effects of global warming and resource scarcity, the sustainable development of furniture is receiving increasing attention. Previous studies have shown that the furniture industry accounts for a substantial portion of global trade [1], uses a large number of raw materials, emits exhaust gases, and generates a large amount of waste. Millions of products are produced, consumed, and disposed of every day, leading to resource depletion and waste generation [2]. These unsustainable patterns of production and consumption form the basis of the current linear (acquire-make-waste) economy [3]. From this point of view, the transformation of furniture production and consumption is particularly important for environmental protection and social sustainable development.



Citation: Liu, J.; Liu, Z.; Yang, Q.; Osmani, M.; Demian, P. A Conceptual Blockchain Enhanced Information Model of Product Service Systems Framework for Sustainable Furniture. *Buildings* **2023**, *13*, 85. https://doi.org/10.3390/ buildings13010085

Academic Editors: Muhammad Shafique and Saeed Banihashemi

Received: 4 November 2022 Revised: 10 December 2022 Accepted: 19 December 2022 Published: 29 December 2022



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Product–service systems (PSSs) have been recognized as "changing production and consumption patterns that may accelerate the shift towards more sustainable practices and societies" [4], and are often outlined as one of the potential enablers for allocating new business models for the circular economy (CE), with the potential to facilitate society's transition to a CE and sustainability [5]. Meanwhile, PSSs can enable sustainable value creation among social, economic, and environmental stakeholders [3]. Currently, the PSSs has been widely used in various fields, such as clothing rental [6], construction [7], and consumer electronics [8], to address growing environmental and sustainability concerns.

In the whole lifecycle of furniture, involving the stages of product design, raw material procurement, and furniture production and distribution, as well as recycling and disposal, it is inevitable to generate a large amount of data and information. These data and information resources are generated by different cross-organizational stakeholders, such as the mutual communication between designers, producers, transporters, and assemblers [7]. Meanwhile, the product itself generates a lot of data and information, such as information about different product components. However, there are some challenges in the process of transforming various types of data into value and ultimately delivering it to stakeholders through various tools [9], and examples include critical data breaches and losses, untraceable logistics, and customer privacy and trust issues [10]. Therefore, it is essential to have an appropriate information model (IM) to guide information collection, sharing, and management [11,12] to ensure secure integration and exchange of information among stakeholders in PSSs lifecycle management.

As digitization becomes a major trend and the role of digital technologies in industry is highly valued, increasing digitization is making possible the vision of a CE of regeneration and recovery [13,14], especially including Industry 4.0 technologies, such as artificial intelligence, big data analytics, cloud computing, Internet of Things (loT), and blockchain (BC) [15,16]. A key goal and feature of Industry 4.0 is to facilitate real-time interconnection between machines, digital devices, transactions, and various stakeholders including suppliers, buyers, and customers [16]. BC technology has the potential to underpin and bring security to the Industry 4.0 vision, which promises to make this possible in a more secure way [13].

BC is considered to have transformative power to change everything from the way business operates to driving the global economy [17], and it has the potential to develop the sustainable furniture industry. Its inherent immutability, transparency, and way of redefining trust relationships by providing fast and secure solutions that can operate publicly or privately have key advantages [18]. Transactions between different stakeholders can be recorded and updated simultaneously and in real time [19]. According to previous studies, BC technology has been widely applied in various fields to cope with various risks, such as processing and protecting patient medical records [20], reducing forged documents and verifying their accuracy [21], and identity management and document authentication in e-government [22].

While both PSSs and BC technologies have been applied to the furniture industry to address related challenges—for example, PSSs have been applied to small furniture manufacturing companies [23], public procurement of refurbished furniture [24], and office furniture manufacturers [25]—BC technology has been applied in furniture mass customization [26], furniture manufacturer value chain [27], and teak supply chain trace-ability system [28].

However, few studies have established the connection between BC, PSSs IM, and sustainable furniture, and limited literature has applied PSSs and BC technology to the whole lifecycle of furniture to promote its sustainable development. Annarelli et al. (2016) found that only a few studies in the academic literature on PSSs have addressed sustainability as a major theme [29]. Meanwhile, as noted by Mont (2002), there are some fragmented PSSs solutions, but few examples are done on a lifecycle basis [4]. Therefore, this paper aims to integrate the potential roles of BC and PSSs in the lifecycle of sustainable furniture to narrow this limitation. It is the first attempt to construct a conceptual BC-enhanced PSSs IM (BC-PSSs) framework for sustainable furniture from the perspective of lifecycle stages.

The rest of this article is organized as follows: Section 2 introduces the literature review of this article; Section 3 describes the research methods of this paper; Section 4 explores the connection between furniture, PSSs, and BC in the quantitative analysis of the literature. In the qualitative analysis, the lifecycle stages of sustainable furniture in this study are clarified, and the PSSs and BC are integrated into sustainable furniture to solve related challenges. Section 5 establishes a conceptual framework for sustainable furniture based on a BC-enhanced PSSs IM; Section 6 reviews and refines the conceptual framework; Section 7 discusses the contribution of the BC-PSSs conceptual framework in sustainable furniture and the potential of BC-PSSs in promoting the value contribution of designers and promoting sustainable behavior by consumers; Section 8 summarizes this article.

2. Literature Review

2.1. Furniture

Furniture usually contains a wide range of products used in everyday life in domestic and non-domestic spaces for storage, lying, support, sitting, working, hanging, and eating functions. Typical products are desks and tables, chairs, cabinets and closets, beds, and sofas [30]. Furniture is closely related to people's life, study, work, and entertainment. The function, beauty, safety, and comfort of furniture are all needed by people. In most industrialized countries, the furniture industry is a basic industry, accounting for 2% to 4% of the manufacturing output value [30]. According to the latest report by Zion Market Research, the global furniture market was valued at approximately USD 331.21 billion in 2017 and is expected to reach approximately USD 472.3 billion by 2024, growing at a compound annual growth rate (CAGR) of approximately 5.2% between 2018 and 2024 [31].

The furniture industry is essentially an assembly industry, which uses a variety of raw materials to product its products [32]. Therefore, the environmental impact brought by furniture is mainly defined by materials, and previous studies have shown that onethird of all materials extracted from the earth are used in the furniture industry [33]. The main raw material used in furniture is wood, which offers various benefits such as high insulation capacity, renewable resource and fuel, and low weight. Other materials, such as chemicals, plastics, metals, and textiles, are also used in furniture [34]. The furniture industry is one of the markets most affected by consumer behavior [35]. Previous studies have also emphasized the link between user behavior and the durability of furniture products, believing that improper use and excessive use will lead to shorter product life [36]. However, the current trend of consumer behavior seems to be to shorten the service cycle of products, especially the use of furniture in business, as most furniture will not be used before becoming old or damaged, and due to aesthetic reasons and image changes, old furniture will be replaced by new furniture [37]. As a result, the furniture industry has a serious waste problem, which in 2017 amounted to the equivalent of 10.78 million tons of furniture waste per year in the EU-28 countries, accounting for more than 4% of total municipal solid waste (MSW). In addition, 80% to 90% of EU furniture waste in MSW is incinerated or sent to landfills [38]. It is estimated that approximately 12 million pieces of office furniture are disposed of as waste each year in Germany [39]. The linear flow of materials and energy leads to the depletion of natural resources and the generation of large amounts of waste [40].

Throughout the lifecycle, including the purchase of raw materials, manufacturing, and waste disposal, each link transfers matter or energy to the natural environment. According to the 17 UN Sustainable Development Goals (SDGs) [41], the furniture industry violates SDGs 11, Sustainable Cities and Communities (Goal 11 target 6 "by 2030, reduce the adverse per capital environmental impact of cities, including by paying special attention to air quality and municipal and other waste management"). It also violates SDGs 12, responsible consumption and production (Goal 12 target 3 "by 2030, achieve the sustainable management and efficient use of natural resources"). This demonstrates the significance of

adopting sustainable practices to reduce the environmental damage and social impact of corporate production [32]. The CE is seen as a significant driver of sustainable development, as it can prevent resource depletion, recycle maintenance materials and products, and recycle potential waste [42]. Therefore, the sustainable development of the furniture industry in the current context is crucial.

2.2. Product–Service Systems (PSSs)

The term PSSs was coined in the late 1990s. Its concept has now been discussed for more than a decade because of its role in sustainable development [3,29,43], and more recently on its contribution to CE [44,45]. As the name suggests, a PSSs consists of three key elements: product (P), service (S), and system (S). In general, products are considered tangible and marketable; Services are intangible activities that are carried out without the need for tangible goods; a system is a collection of products, services, and their relationships [46].

Over the years, different researchers have adopted different terms for the concept of the PSSs. Generally, from the definition of Goedkoop et al. (1999) "a combination of products and services in a system that provides functionality to consumers and reduces environmental impact" [46]. This shows that the initial focus of the PSSs was on environmental sustainability. Currently, Annarelli's latest definition "PSSs is a business model focused on providing a set of marketable products and services aimed at achieving economic, social and environmental sustainability, with the ultimate goal of meeting the needs of our customers" [29]. This definition includes all aspects of sustainability, indicating that the original potential for achieving sustainability through a PSSs is being refocused.

Although researchers have different views on the definition of a PSSs, they mostly adopt the same classification, which divides PSSs into three categories according to the degree of dematerialization [47,48]. (1) Product-oriented PSSs: the transfer of ownership of the physical product to the customer to provide additional services (such as repair and recycling services) for the product to improve the life or utility of the product. (2) Use-oriented PSSs: where the ownership of the physical product is owned by the manufacturer, and the use and functions of the product are provided to the customer. (3) Results-oriented PSSs: Ownership of the physical product is retained by the manufacturer, who provides the result or capability to the customer, a process that involves a substantial system shift in which the physical product is primarily replaced by the service. These strategies can help foster CE and promote sustainable development. For example, product-oriented PSSs can extend the product lifecycle, such as maintenance, and use-oriented PSSs can help companies deal with product use processes, such as leasing [49]. Therefore, a PSSs has the potential to promote sustainable production and consumption and support the transition to CE.

Lifecycle analysis has always been an important topic in PSSs [50]. The lifecycle of a PSSs refers to the whole process from the formation of design concept to the end of use, which involves various information exchanges among many stakeholders. To meet this requirement, providing integrated information management related to products and services throughout the lifecycle is essential [51]. IM has been developed in this area to efficiently collect, exchange, and manage information. For example, using lifecycle information to improve PSSs products and related activities [52], integrating information and data to support PSSs value co-creation [53], and realizing closed-loop lifecycle information sharing in a PSSs [54]. The establishment of PSSs IM has the potential to break vertical information islands in traditional business models [54], coordinate discrete stakeholders [55], and facilitate information exchange at different stages of the life cycle [56].

PSSs lead to a new understanding of the development strategy and value goals of manufacturing enterprises. This business model, whether in a business-to-business (B2B) or business-to-consumer (B2C) environment, integrates products and services to provide value [57]. PSSs are also a potential tool to support social sustainability [58] and the transition to CE by changing companies' incentives to extend the life of their products, thus making them more cost effective and resource efficient [59].

2.3. Blockchain (BC)

Recently, with the continuous emergence of digital technologies, such as BC, cloud computing, big data, and artificial intelligence, as well as the deep integration with various industries, the digital economy has penetrated into every link of economic and social development [60,61]. As a new economic form with digital knowledge and information as key production factors, data privacy, and data security have become the focus of high-quality development of the digital economy [62]. BC technology is suitable for the digital economy in addressing privacy and data security issues and is becoming an important technological foundation for the development of the digital economy [63].

The essence of BC is a series of blocks with records and time stamps, connected by hash values, which has been hailed as foundational [19] and disruptive technology (Jesse, 2018). The key attractive features of BC technology include distributed databases, peer-to-peer transactions, transaction automation, anonymity and transparency, and irreversibility of records.

- 1. Distributed database. A BC is described as "a distributed database that is agreed upon and shared across peer-to-peer networks" [64]. Each node in the system stores all data, which means that there is no centralized database [65]. This makes the BC a very secure distributed ledger, so all parties have access to the same and correct records and the complete history of the entire database [19,66].
- 2. Peer-to-peer transaction. BC allows for the sharing of records with all network nodes without central authority; peer-to-peer interconnected ledgers ensure that updates to any one ledger are visible and accessible to all [67]. Decentralization is key to promoting trust and security among interconnected parties, thereby facilitating the digitization of business processes [68]. Through this platform, stakeholders can access the same data and transactions without a single authority and any geographical restrictions [67]. This can enable all components (things) to trade autonomously without the intervention of regulators or brokers [68].
- 3. Trading automation. BC technology can be used to automate transactions, which can be carried out through digital, automated "Smart contracts" [19,66]. Smart contracts are self-executing snippets of code defined by pre-specified terms that are publicly visible to all nodes on the network or BC, and the smart contract will be triggered when the preset obligations are met [69,70]. Meanwhile, the diversified implementation of smart contracts in different fields is also expected to reduce transaction costs [13].
- 4. As a repository of transaction lists recorded by forgery-proof documents, BC has the characteristics of application security, reliability, transparency and automation [71]. It is designed to ensure security and use cryptography and distributed consensus mechanisms to provide anonymity, persistence, auditability, resilience and fault tolerance [72]. With its open and transparent nature, BC provides a trusted environment for distributed companies using the Internet for business and transactions, which facilitates record and operations, as well as stakeholder acceptance and satisfaction [67].
- 5. Irreversibility of records. Security can be further enhanced by using encryption algorithms. All data stored on the BC have a timestamp and asymmetric encryption attached to the BC, which is mainly used to protect ledger consistency and irreversibility of records, ensuring that block data cannot be falsified and tampered with [71]. Once a transaction is updated and verified by all relevant nodes or key parties in the network, it is irreversible and cannot be rewritten or reordered, which ensures data security [65].

In addition, BC has the potential to support the transition to a CE and promote sustainable development. Sustainability and broader social responsibility are achieved through business models that promote circular value creation of decentralized products and services [13]. CE aims to release environmental pressure from economic enhancement and coordinate the relationship between environment, economy, and society [73]. With its

composition of open source, peer-to-peer, and distributed ledger systems and automation capabilities, BC has the potential to create cleaner economic transaction processes and help achieve much needed balance and harmony between the environment, economy, and society [13]. BC is seen as a socio-technical or coordination tool that helps connect and coordinate multiple distributed databases [74]. By recording critical data in the BC, long and complex supply chains can be monitored with relative ease and efficiency in the process of products from raw material source to manufacturer to customer [75]. Meanwhile, it can irreversibly update all connected databases and facilitate automation where needed [13]. Therefore, through BC cooperation and knowledge sharing, sustainable development and social responsibility goals can be achieved and CE can be promoted.

3. Research Method

In this paper, a conceptual BC-enhanced PSSs IM framework for sustainable was constructed based on a combination of quantitative and qualitative literature analysis and verified and improved through questionnaires and interviews with industry experts and scholars.

In the quantitative analysis, VOSviewer bibliometrics software was used for analysis, which could reflect the frequency of keywords, the relationship between keywords and the cluster form of keywords by forming the keyword co-occurrence visualization network [76,77]. Therefore, this paper uses VOSviewer software to analyze the relationship between furniture, the PSSs, and BC. In this process, the relationship between the PSSs and furniture sustainability and CE is explored first. Then, in the context of digitization and digital economy, the relationship between BC and the PSSs is discussed. According to the analysis of the previous two, the promise of BC and the PSSs in sustainable furniture is mined.

Qualitative analysis was carried out based on the quantitative analysis results. Firstly, the lifecycle stages of sustainable furniture in this study are determined. Secondly, the PSSs is applied to sustainable furniture and the advantages are analyzed. Then, around the whole lifecycle of sustainable furniture, we summarize the current challenges of sustainable furniture PSSs. Finally, we analyze the advantages that BC brings to the PSSs IM framework and integrate BC into PSSs IM to address the challenges related to sustainable furniture. On the basis of quantitative and qualitative analysis, a conceptual BC-enhanced PSSs IM (BC-PSSs) framework is constructed for sustainable furniture, and BC is used to address the challenges related to the lifecycle of sustainable furniture. The conceptual BC-PSSs framework was reviewed by industry experts and scholars in the "Chinese Society of Industrial and Applied Mathematics BC Technology and Application Summit Forum 2021" (CSIAM-BTAF 2021) [78], which helps to further refine the conceptual BC-PSSs framework. The overall structure of this paper is shown in Figure 1.

The conceptual BC-PSSs framework was validated in the form of pre-interview questionnaires and interviews. SPSS 22 software was used to examine the reliability of quantitative questionnaire data, and NVivo 11 software was used to examine the content of qualitative interview data. To conduct rigorous qualitative research, Lincoln and Guba created reference standards including credibility, reliability, confirmability, and transferability [79,80]. Therefore, the criteria were adjusted according to the research strategy of this paper, as shown in Table 1.



Figure 1. The flow chart of the research methodology (generated by authors).

Criteria	Purpose	Strategies Applied in This Study
Credibility	To believe that the results (from the view of the participants) are credible, true, and believable.	Ensure that investigators have the knowledge and research skills required to perform their duties.
		A detailed draft study protocol was prepared throughout
		the study.
Dependability	To ensure that the findings of qualitative investigations are reproducible.	A detailed track record of the data collection process was developed.
		Measure the coding accuracy and reliability of the
		research team.
Confirmability	To establish confidence that the results will be	Apply multiple techniques (methodology, data sources,
	confirmed by other researchers.	investigators, and theory).
Transferability	To extend the degree to which the results can	Sampling purposefully to form a specified sample.
manorerability	be generalized to other contexts or settings.	Quantified interview transcripts.

Table 1. Strategies adapted from Lincoln and Guba applied in this study.

4. Quantitative and Qualitative Analysis

The VOSviewer software was used to analyze the relationship among furniture, the PSSs, and BC. First, we explore the potential of the PSSs to enhance the sustainability of furniture and promote CE. Secondly, in the context of digitalization and digital economy, the link between BC technology and PSSs is discussed. Then, we integrate BC and PSSs to bring the promise of sustainable furniture.

4.1. Quantitative Analysis

4.1.1. Furniture and PSSs

This study aims to understand the current research on furniture and PSSs in developing CE and sustainability and to explore the potential connection between furniture and PSSs. Retrieve TS = (furniture AND circular economy), TS = (furniture AND sustainability), TS = (product-service systems AND circular economy), and TS = (product-service systemsAND sustainability) were searched for in the core collection of Web of Science, and 53, 242, 227, and 540 articles were found without the restriction of year, respectively. All the retrieved documents were imported into VOSviewer software for keyword co-occurrence analysis, and the network visualization diagram was generated, as shown in Figure 2. This figure is a visualization of the overall network, which mainly shows the cluster of existing keywords in the search field, as well as the relationship between each keyword. According to Figure 2, the high-frequency keywords (frequency > 100) in this field are mainly "sustainability", "circular economy", "design", "product-service systems", "innovation", and "management". Meanwhile, in the context of CE and sustainability, the keywords closely related to "furniture" include "wood", "waste", "management", "behavior", "products", and "attitudes"; additionally, "product-service systems" is mainly closely related to "circular economy", "business model", "consumption", "economy", and "environmental sustainability" and "barriers". To further explore the connection between furniture and PSSs, the opportunity to establish the relationship between them is sought in the overall network visualization diagram. Through analysis, it is found that "lifecycle assessment" serves as a bridge to connect "furniture" with "product-service system", as shown in Figure 3 (Figure 3 is the highlighted map of the overall network visualization in Figure 2).



Figure 2. Keywords co-occurrence network visualization of furniture and PSSs in the context of circular economy (CE) and sustainability (generated by authors).



Figure 3. Keywords co-occurrence visualization of furniture and PSSs in the context of CE and sustainability highlighted from the Figure 2 (generated by authors).

4.1.2. PSSs and BC

This study aims to understand the current research of PSSs and BC in the development of digital economy and digitalization to explore the potential connection between PSSs and BC. The authors searched for TS = (product-service systems AND digital economy),TS = (product-service systems AND digitization), TS = (blockchain AND digital economy), and TS = (blockchain AND digitization) in the core of Web of Science, and 40, 43, 272, and 143 were found without the year limit, respectively. All the retrieved documents were imported into VOSviewer software for keyword visualization analysis, and keyword network visualization diagram was obtained. Figure 4 mainly shows the clustering of existing keywords in this retrieval field, as well as the relationship between each keyword. According to Figure 4, the high-frequency keywords (frequency > 35) studied by PSSs and BC in the context of digital economy and digitalization are mainly "blockchain", "digitization", "Internet", "technology", and "blockchain technology". Meanwhile, in the context of the digital economy and digitalization, "blockchain" is mainly closely related to "digital economy", "blockchain technology", "bitcoin", "technology", "big data", "cryptocurrency" and "innovation"; The keywords closely related to "product-service systems" are mainly "servitization", "transition", "digitization", "management", "systems", and "Industry 4.0". According to Figure 5, "digitization" acts as a bridge to connect "blockchain" and "blockchain technology" to "product-service systems".



Figure 4. Keywords co-occurrence network visualization of PSSs and BC in the context of digitization and digital economy (generated by authors).



Figure 5. Keywords co-occurrence network visualization of PSSs and BC in the context of digitization and digital economy (detail map) (generated by authors).

4.1.3. Furniture, PSSs, and BC

The relationships between furniture and PSSs and PSSs and BC are discussed, respectively. Further discover the potential promise of PSSs in promoting sustainability and CE in the furniture industry, as well as the potential possibilities of BC in promoting PSSs. This underpins the idea of using BC to enhance PSSs for sustainable furniture in this study. Meanwhile, according to the core collection of Web of Science, research on "furniture AND product–service systems AND blockchain" is still lacking, so this research can fill in the relevant research gaps to some extent.

4.2. Qualitative Analysis

According to the quantitative analysis results of relevant keywords by VOSviewer software, the following research contents are qualitatively analyzed. Firstly, the lifecycle stage of furniture in this study was determined. Then, this study applies PSSs IM to the lifecycle of sustainable furniture, analyzes the advantages of PSSs for sustainable furniture, and summarizes the challenges faced by sustainable furniture PSSs from the perspective of furniture lifecycle. Then, it analyzes the advantages of BC in PSSs IM and uses BC to overcome the challenges faced by PSSs IM in sustainable furniture. It lays a theoretical foundation for constructing the conceptual BC-PSSs framework for sustainable furniture.

4.2.1. The Lifecycle Stage of the Furniture

The lifecycle of a resource usually consists of four successive phases: source, production, use, and end of lifecycle [57]. In a product, the lifecycle can be understood as a series of events that the product goes through, including not only the actual production, but also the services that accompany the whole lifecycle of the product [65]. Meanwhile, environmental impacts run through the entire product lifecycle, from the extraction of natural resources as raw materials, the production process, the use and reuse of products, and finally the disposal stage [81]. According to the product category rules of furniture [82], the lifecycle is divided into upstream stages (processes related to raw materials, manufacturing processes, energy use, and waste generation during production), core stages (processes related to transportation, product assembly/manufacturing, energy consumption, and waste disposal during manufacturing), and downstream phases (processes related to the retailer/distribution platform, transportation, product maintenance, product use, and obsolescence process). In previous studies, researchers have different definitions of the lifecycle stages of furniture based on different research focuses, as shown in Table 2.

Research Focuses	Items	Resources
The process of furniture industry	Raw material supply, manufacture, distribution, use, and disposal	Besch, 2005 [34]
Furniture design assessment	Design stage assessment, production stage assessment, use stage assessment, sales stage assessment, and recovery stage assessment	Chen et al., 2017 [83]
Wood furniture lifecycle	Design, manufacture, manufacture/distribution, use/maintenance, and end of life	Bianco et al., 2021 [82]
Furniture lifecycle (system perspective)	Upstream activities (i.e., production, supply, and processing of materials and components), core activities (i.e., product manufacturing, assembly, finishing, packing, and storage); and downstream activities (i.e., product distribution, retail, use, maintenance, and end of life).	Cordella & Hidalgo, 2016 [30]
The lifecycle of furniture products	Production and supply of materials (P1), product manufacturing (P2), distribution (P3), use and maintenance (P4), and end of life (P5).	Cordella & Hidalgo, 2016 [30]
Furniture lifecycle	manufacturing, retail/distribution, use/maintenance, and end of life	Bianco et al., 2021 [82]

Table 2. Previous research related to the definition of furniture lifecycle stages (generated by authors).

In this study, the lifecycle stages of sustainable furniture are defined as eight stages, including design, procurement, manufacture, transportation, distribution, maintenance, recycle, and disposal stages, which are described as follows:

- 1. Design stage: The lifecycle begins with conceptual design, which involves information collection, ideas, design drawings, computer-aided design, innovation, and development of new products, etc. Decisions made at this stage are very important because they have an impact on the whole lifecycle, from manufacturing to the end of life [84]. This is because the binding of product materials and production techniques chosen during the product design phase will determine the pollutants and wastes released and energy consumed during its use, as well as the ease with which its components can be reused in subsequent use and manufacturing cycles [85,86]. It is estimated that 70% of the environmental impacts produced by products are determined at the design stage [87]. Therefore, most of the sustainability characteristics of a product are attributed to the early design phase [88].
- Procurement stage: Procurement is a prerequisite activity to ensure clear production, which involves inventory management, raw material procurement, etc. The determination of procurement and inventory is very important to minimize inventory costs [89], because the supply of raw materials will affect production and thus sales.
- 3. Production stage: Production is the process of transforming design concepts into tangible objects. The production of furniture basically includes the assembly of components and the surface treatment of products, which involves the production of component status, production planning, production execution, and production investment to ensure the smooth progress of the production process. The environmental burden at this stage is mainly due to the consumption of electricity and heat energy, but also the consumption caused by the use of painting, drying, adhesives, solvents, and some chemicals [30]. Optimizing resource utilization may be the most effective measure to reduce the environmental impact of furniture production, as significant environmental benefits can be achieved by improving the energy efficiency of the manufacturing process and increasing the use of renewable energy, as well as recycling materials and waste [90].
- 4. Transportation stage: Transportation delivers products to retailers, which involves container and logistics management, etc. The impact of this stage can be reduced by using more effective transportation methods and optimizing loading and logistics

strategies [90]. For example, giving priority to suppliers close to the furniture production site may be a relevant measure to reduce the environmental burden caused by the transportation process.

- 5. Distribution stage: distribution sells products to consumers, which involves product display, product transportation, and contract signing.
- 6. Maintenance stage: The durability and actual use time of the products will greatly affect the impact of furniture products on the environment. Maintenance can extend the lifecycle of a product, and products that are usually easy to clean and dismantle/repair can reduce the environmental impact of this stage. In particular, the effectiveness of any action in this area itself depends on consumer behavior [30].
- 7. Recycle stage: The option of recovering value from furniture products after their use is widely deployed. Promoting recycling systems can also serve as additional support to the options listed, as well as minimizing the amount of materials and components used in products and promoting the use of easily identifiable and separated recyclable materials and reusable components [91].
- 8. Disposal stage: Disposal is the end of the life of furniture, which may have a significant impact on the determination of product environmental conditions. The contribution variation at this stage is basically based on the disposal option considered; for example, landfill is the worst disposal option, whereas reuse of products or product components can directly avoid the environmental impacts associated with the production of new units [30].

4.2.2. Lifecycle Management of PSSs Information Model (IM) in Sustainable Furniture

"PSSs is a system of products, services, support networks and infrastructure designed to be competitive and meet customer needs with a lower environmental impact than traditional business models", according to Mont's (2001) overview, a PSSs means not only the services and products delivered, but also the infrastructure and networks that facilitate the system [92]. This fact supports the claim that PSSs development requires interactions across the entire product chain [93]. Therefore, a broad view of the product–service lifecycle is critical to achieving good PSSs performance.

The PSSs aims to maximize the use of the product and its maintenance, prioritizing system administration throughout the product lifecycle [43]. According to the research [34], the establishment of a PSSs needs to include three important roles, namely the manufacturer, retailer, and consumer. Manufacturers are groups that focus on the design, production, and re-manufacturing of furniture; the retailer is the link between the manufacturer and the customer. Many furniture manufacturers sell their products through the retailer, and the customer buys the products through the retailer and signs the full service contract with the retailer. Given the key role of consumers in CE [45], in addition to production and distribution, the whole circular supply chain should also consider the consumption process; particularly, the level of consumer awareness of environmental and sustainability issues may greatly influence the practical implementation of circular business models [94].

Based on the above overview, PSSs IM is defined in this paper. Firstly, according to the three important roles of manufacturer, retailer, and consumer in the PSSs proposed by Besch (2005), this paper defines the manufacturer as the manufacturing group of products, the retailer as the distribution group, and the consumer as the consumption group [34]. Meanwhile, according to Section 2.2 of this paper and the three key elements of product, service, and system in the PSSs, in this paper, the product corresponds to the manufacturing group, the service corresponds to the consumer group, and the retailer is the link between the product and service, reflecting the transmission of product to service, and finally forming a complete system combining product and service. This is shown in Figure 6.



Figure 6. The composition of the product–service systems (PSSs) information model (IM) in this study (generated by authors).

According to the lifecycle stage of sustainable furniture, the specific embodiment of PSSs IM applied to sustainable furniture is shown in Figure 7. "Product" in the PSSs corresponding to the "design stage", "procurement stage", and "manufacture stage" of sustainable furniture belongs to "manufacturing groups". The "transportation stage" and "distraction stage" of sustainable furniture correspond to the "product–service" in the PSSs, which are "distraction groups". The "maintenance stage", "recycle stage", and "disposal stage" of sustainable furniture correspond to "service" in the PSSs, which belongs to "consumer groups". Finally, the whole lifecycle of sustainable furniture forms a system that combines products and services.



Figure 7. Sustainable furniture PSSs IM in this study (generated by authors).

According to the potential of a PSSs in facilitating stakeholder assistance and enhancing lifecycle knowledge management, in this paper, PSSs is used for the stakeholders and lifecycle stages of sustainable furniture to play its potential role. Figure 8 illustrates the ability of PSSs IM to enhance sustainable furniture. With the introduction of PSSs, the coordination and communication between the key stakeholders in sustainable furniture such as designers, suppliers, engineers, and producers is further enhanced. At the same time, the sustainable furniture lifecycle phase closes the material and energy flow, reduces the exploitation of natural resources, and reduces waste generation, and the material resources in the end of the lifecycle process will be re-channeled to the beginning of the next product lifecycle. Various information from stakeholders and lifecycle stages flow with the support of the PSSs, facilitating coordination and improving collaboration and communication, while lifecycle knowledge management is carried out to validate, manage, and share all information assets of the organization, such as databases, documents, and organizational policies and procedures, as well as to potentially harvest the tacit skills and experience of knowledge workers. In addition, the PSSs embodies the core concept of CE "reduce, reuse, and recycle" throughout the lifecycle of sustainable furniture, in line with the goal of a CE to break away from the linear approach of "make, use, dispose" and promote the sustainable development of the furniture industry.



Figure 8. The lifecycle of sustainable furniture enhanced by PSSs IM (generated by authors).

4.2.3. PSSs Advantages for Sustainable Furniture

The economic and ecological potential of the PSSs lies in the area of optimization of the whole system, extending from a single product lifecycle to an integrated product–service lifecycle [49,95]. From a systems point of view, products are designed not only to provide innovative learning environments that meet current needs, but also to support the sustainable development of society through sustainable production and consumption and CE. In other words, the PSSs is not only a collection of products, services, and other elements, but also includes the value proposition, the type of interaction between stakeholders, and the solution to social problems [3]. Systematic lifecycle management of PSSs brings the following benefits to sustainable furniture.

- 1. Lifecycle knowledge management. Knowledge management is the organizational and technical basis for enterprises to promote knowledge sharing and reuse [96]. PSSs can reasonably and effectively manage the products, services, data, and information generated in the whole lifecycle, and provide the required knowledge to different stakeholders in each lifecycle stage [97].
- 2. Forming a closed loop of materials. The PSSs minimizes the impact of consumption on the environment by closing the material cycle, including the return of furniture from the customer to the manufacturer, in which case furniture elements can be remanufactured, updated, and reused [34]. One can get the most out of the product and its maintenance through lifecycle system management.

- 3. Promote collaboration among stakeholders. The PSSs enables networked stakeholders in the product value chain to collaborate effectively [7]. These centers are based on strategic partnerships between manufacturers and suppliers, customers, supporting institutions, and other stakeholders to promote the development of a culture of value-added products in the furniture industry [98].
- 4. Improve the competitiveness of enterprises. The PSSs can strengthen the relationship between enterprises and consumers, and the construction of this relationship promotes the loyalty among consumers [99]. In addition, companies can use the information gained from consumer relationships to develop new systems to improve product performance [100]. Companies can improve their position in the value chain and increase their innovation potential [101].
- 5. Increase resource productivity. Servitization offers access to a functional dematerialized future that is possible in the long run, as well as opportunities to increase resource productivity in the short run [48]. Increased service supply compensates for job losses due to cutbacks in traditional manufacturing and relieves consumers of responsibility for services, such as installation and maintenance, as well as product disposal at the end of their useful lives [43].

4.2.4. The Challenges for PSSs in Sustainable Furniture

Although PSSs have great advantages in the lifecycle of sustainable furniture, there are still some challenges that cannot be solved. The following are eight major challenges to sustainable furniture PSSs from a lifecycle perspective.

- 1. The intellectual property rights of creators cannot be protected. One of the characteristics of product design is the interdisciplinary nature of the design process, which means that designers may work with material engineers, structural engineers, and other professionals [102]. The results of these professionals' work together promote the development of new products. However, with the transfer of work results, the knowledge transfer becomes difficult to track, and the legal rights of the knowledge contributors cannot be ensured. At the same time, unauthorized copying, "stealing", and other infringement methods have become common scenes, which also seriously damage the rights of original creators [103]. Existing central website-based copyright management where the content is inaccurate, easy to attack, without strict and credible traceability, and other shortcomings need to be addressed [71].
- 2. Improper inventory management of raw material procurement This includes lack of trust and transparency among key stakeholders in the procurement process, weak system support for transaction records and documentation, and complex process structures [104]. Due to the improper delivery of raw materials by suppliers, the shortage or unavailability of raw materials occurs from time to time. This will lead to serious consequences: First of all, the distribution of raw materials is delayed and the production schedule will be disrupted, which leads to delays in receiving the goods by customers, which affects consumer satisfaction. Secondly, prompting enterprises to rush to order from other suppliers, the result of urgent orders is the emergence of additional costs for companies [105].
- 3. Product elements lack standardized specifications and production principles. Due to the lack of a standardized production system, the production process is characterized by low production efficiency and long and unstable production cycles. Furniture products include different manufacturing materials, different colors and specifications, a large number of various types of parts, large differences in process production, and lack of standardized specifications and production principles, which will cause management difficulties and affect delivery [106].
- 4. Lack of logistics planning and low transport efficiency. Efficiency, speed, and agility in transportation operations are essential elements that make up the furniture manufacturers of the future. However, even if different furniture companies located in the same area ship to the same market area, the same city, and/or the same furniture

retailer, coordination of transportation operations between two or more companies is rare [107]. One reason for the inefficiency is that most trucks return to the plant empty. The environmental impact caused by product transportation is the main environmental burden of the office furniture industry [108].

- 5. Product distribution brings distrust to consumers. More and more consumers are aware of environmental concerns and therefore tend to buy sustainable products. However, certain green advertisements present a mixture of facts, ambiguous information, and provide misleading and exaggerated information about the green benefits or environmental or health attributes of the product [109]. These lack of credibility advertisements hinder consumers' trust in green furniture and give users a poor or even unbearable experience. Some consumers are skeptical of eco-labeled products, have a lack of trust in the products being sold, and have a genuine commitment to trust attributes that may reduce consumers' purchase intention [110]. In addition to this, certain low-quality ads may contain viruses that induce users to click and then steal private data [111].
- 6. Maintenance service cannot be effectively guaranteed. On average, consumers buy new furniture every three years. The main reason for consumers to buy new furniture is the need to replace worn or damaged furniture [101]. The biggest problem found after three years of use of furniture is the wear of parts, material deterioration, and furniture companies that do not support repairs and do not provide spare parts services to customers. As a result, accumulated end-of-life products are discarded and difficult to manage, eliminate, and destroy [112].
- 7. Lack of a complete recycling system. Most waste furniture is burned or buried [113], resources are not effectively utilized, and even the environment is affected by waste treatment means [114].
- 8. Lack of secure data storage and transmission. The furniture lifecycle includes multiple stages, and various data are gradually detailed through each stage; as a result, there are a large amount of data. Various data and knowledge resources are held by different cross-organizational stakeholders, such as manufacturers with status data and detailed design, manufacturing and reliability knowledge, service providers with maintenance data and experience knowledge, and consumers with usage data for systems [115]. Nevertheless, the PSSs can facilitate the collection, management, and sharing of resources. However, it is difficult to ensure the security and privacy of shared knowledge, and the loss and destruction of data can greatly reduce the effective transmission of information.

4.2.5. The BC Potential Advantages for PSSs IM

According to Section 4.1.2, there is a potential connection between BC and PSSs. Through the relevant literature, it is found that BC has potential advantages in PSSs IM. The PSSs is seen as a novel business model that focuses on cost, convenience, CE, and the environment [116], and it is able to improve value creation by improving circularity [117]. The value creation part of the business model is used in scenarios where products or services are provided to customers. However, in order to reasonably and effectively manage the products, services, information, and data generated throughout the lifecycle of a PSSs when providing services throughout the lifecycle of a product, consideration needs to be given to how to securely track and trace products, how to trigger services, and how services are delivered, and the problem of data insecurity and low trust among stakeholders must also be overcome [65].

BC can help stakeholders in PSSs store data in a more secure way, enabling decentralized peer-to-peer transactions that are anonymous and visible to everyone in the chain [65]. This provides strong support for later data utilization and service innovation, providing new ways to improve traceability and transparency throughout the product lifecycle [118]. Specifically, the lifecycle of PSSs refers to the entire process of PSSs from the formation of the design concept to the final use, which contains a number of different stages, and the BC can improve the traditional PSSs lifecycle management from multiple aspects [97]. For example, in the creative and design phases, BC enables companies to apply recommendation algorithms, where manufacturers can get accurate recommendations without the customer revealing his/her data; during the implementation phase of products and services, the potential benefits of BC include trust among stakeholders, greater supply chain efficiency, and customer satisfaction. In the use stage of the product, the BC helps the PSSs improve the maintenance and upgrade service, and the irreversibility of the record ensures the efficient and error-free verification of the service results in the PSSs and realizes safe and efficient resource sharing.

4.2.6. The BC Potential to Overcome Challenges for PSSs IM in Sustainable Furniture

BC has the potential to facilitate the implementation of PSSs IM in sustainable furniture. In addressing the eight identified challenges of sustainable furniture, BC has the following potential roles to play:

- 1. Protection of the intellectual property rights of creators. The timestamp and asymmetric encryption features of BC technology ensure that block data cannot be forged and tampered with. For each copyright transaction, such as authorization, distribution, tracking, use, and destruction, the copyright owner will form a unique sub-chain according to time [71]. BC can more securely share expertise and/or business performance data owned by brand/franchise owners, while distributed storage provides transparency and openness to the BC as long as the person registered in the BC is able to know the ownership of the work and other relevant information, thus avoiding the leakage of business-sensitive information, customer records, and patented technology to "third parties" and/or cybercriminals [119].
- 2. Effective inventory management of raw material procurement. BC can contribute to more accurate demand forecasts when planning inventory purchases, not only gaining a cost advantage, but also eliminating unnecessary waste of resources [120]. By integrating BC into furniture PSSs, it can help control and track materials to reduce the likelihood of overstock, defective, and/or illegal inventory [121].
- 3. Improving the standardization and production of product elements. More accurate forecasting of material requirements in the production process reduces the frequency of material expiry due to excess inventory [122]. Each stage of subcontracting takes the form of smart contracts in BC that can be converted into programs and code, which are then copied and stored in a processing system and monitored by a network of systems running the BC [7]. Therefore, the bill of quantities of product elements (PC) can be obtained, and classified management, accurate control of inventory supply and demand, flexible adjustment of PC output, and continuous production capacity of PC can be ensured.
- 4. Effectively implement logistics planning and transportation management. At this stage, BC can be used to prevent security breaches while enhancing transport connectivity and delivery services. Transport is a key player in connecting supply and distribution and enabling a transparent, sustainable, and efficient flow of status data between them [7]. After receiving the order, the delivery plan and driver allocation can be carried out automatically, and the entire transportation phase is also monitored in real time. At the same time, the BC can also store information and images (e.g., component status, driver information, routes), and if an inappropriate operation occurs, the cause and responsible person of such problems can be immediately tracked through the system.
- 5. Improving consumer trust in the distribution process. The use of BC technology has improved the quality and effectiveness of advertising, increased the enthusiasm and experience of users, and improved the adverse impact of the proliferation of low-quality advertising on users [111]. At the same time, BC as a transparent, distributed. and clean register, as well as the use of QR code based on BC technology in the product, can let consumers understand the product source, manufacturer name, and packaging

date, which plays an important role in promoting the traceability of furniture products, facilitating query and not copying [71].

- 6. Improving maintenance services and contract effectiveness. Maintenance and upgrades of the product are especially important because they extend the life of the product and can provide continuous service. With the help of BC, maintenance agencies can create smart contracts based on maintenance standards for different products and submit smart contracts to the BC [97]. When the user submits the information to meet the terms of the smart contract, the maintenance agency can provide repair suggestions or services in a timely manner or even in advance.
- 7. Improving the waste recycling and treatment system. BC can help recyclers verify the status of recycled products and carry out waste tracking and management. BC creates a more efficient and transparent system that helps track waste [123], and waste tracking ensures that recyclables do not appear in landfills [124]. After being properly repaired, waste furniture can be re-marketed as a product, reducing the environmental load [113]. The entire process can be recorded on the BC, and intermediate transactions are done through smart contracts, helping to achieve an efficient and transparent recycling process, enhancing user trust, and promoting sustainable development.
- 8. Ensuring secure storage and efficient transmission of data. The database must maintain data integrity in order to truthfully inform shareholders about progress and any existing issues or obstacles [7]. The use of BC to strictly encrypt the data in sustainable furniture to ensure authenticity not only enhances the value of data sharing in the lifecycle, but also improves the anti-risk ability of PSSs.

According to the above, BC has the potential to facilitate the implementation of PSSs IM in sustainable furniture and solve lifecycle-related challenges, digitally driving the entire lifecycle of sustainable furniture. As shown in Figure 9, the advantages of integrating BC and PSSs IM and applying them to stakeholders and throughout the lifecycle to play a potential role: First, among stakeholders, BC enhances PSSs IM and promotes coordinated trust among stakeholders in sustainable furniture. Secondly, in the lifecycle, BC-enhanced PSSs IM strengthens the links between the various stages of the sustainable furniture lifecycle and solves the associated challenges. BC then enhances the lifecycle knowledge management of sustainable furniture PSSs IM for reviewing and managing knowledge related to different domains to verify, manage, and share all of an organization's information assets such as products, documents, organizational policies, and procedures. Meanwhile, in the whole process, a full lifecycle information database is established with the support of BC, and the information system allows for dynamic marketing communication between enterprise product management, distribution channels, and direct sales personnel, as well as securely storing and transmitting data.



Figure 9. BC-enhanced PSSs IM in sustainable furniture (generated by authors).

5. A Conceptual BC-Enhanced PSSs IM (BC-PSSs) Framework for Sustainable Furniture

The establish of the conceptual BC-PSSs framework for sustainable furniture is based on the results of quantitative and qualitative analysis, as well as the significant lifecycle stages and associated challenges of sustainable furniture. The basic principle of the conceptual BC-PSSs framework is to use BC-enhanced PSSs as a tool to drive sustainability throughout the lifecycle of furniture and to assist stakeholders in interaction and furniture knowledge management. The conceptual BC-PSSs framework consists of a high-level framework, which is strategic, and a low-level framework, which is detailed. The numbers coded in red represent the significant process actions of BC enhance PSSs IM at various stages of the whole lifecycle of sustainable furniture to identify the potential role of BC integrated PSSs IM in the whole lifecycle of sustainable furniture and to meet the challenges of sustainable furniture.

5.1. High-Level BC-PSSs Framework for Sustainable Furniture

Based on the important stages of the sustainable furniture lifecycle, the related challenges, and the advantages of BC-enhanced PSSs IM, the conceptually high-level BC-PSSs framework consists of three layers, including the user layer, system layer, and knowledge layer, as shown in Figure 10.



Figure 10. High-level BC-PSSs framework for sustainable furniture (generated by authors).

At the user layer, BC-PSSs underpins the exchange of information, coordination, and linkages among stakeholders in sustainable furniture (A1); guarantees effective collaboration and trust among stakeholders (A2); and securely stores relevant data (A3).

At the system layer, BC addresses the challenge of sustainable furniture PSSs IM at eight stages of the lifecycle: design, procurement, manufacture, transportation, distribution, maintenance, recycle, and disposal. It can support the collaborative operation of information collection, transmission, and communication, as well as provide a reliable basis for collaboration. In the manufacturing group (B1), BC strengthens trust among professionals (B1-1) and protects the intellectual copyright of creators (B1-2). With the advantage of BC, the procurement process enables more precise planned inventory (B1-3) and smooth production by standardizing production and secure storage of product elements (B1-4). In the distribution group (B2), the BC enables the development of a rational operational plan for the transportation process (B2-1) and the tracking and control of containers or products from loading to real-time delivery (B2-2). In the process of product distribution, the traceability and irreversibility of BC prevent false advertising and publicity of products (B2-3) and improve the transparency of products (B2-4). In the consumer group (B3), BC has the potential to protect the rights and interests of consumers in maintenance services (B3-1) and guarantee the validity of contracts (B3-2), thus extending the lifecycle of products. In addition, it improves the recovery rate of products, from tracking waste (B3-3) and establishing a reliable waste recovery system (B3-4) to the reasonable disposal of various materials and reducing environmental pollution. Throughout the lifecycle, BC ensures the secure storage and efficient transmission of all kinds of data in the design, procurement, manufacture, transportation, distribution, maintenance, recycle, and disposal stages (B1-5, B2-5, B3-5).

At the knowledge layer, BC-PSSs performs knowledge management over the entire lifecycle of sustainable furniture (C1) with the following: knowledge of production and installation of product components (C1-1), knowledge of product materials (C1-2), knowledge of product services (C1-3), and knowledge of product process methods and technical standards (C1-4). The relevant knowledge data are securely stored in the knowledge base (C2), which can provide various collection, maintenance, and retrieval functions for related resources. As a result, relevant personnel can consult their relevant expertise to alleviate the barriers posed by the lack of value information in emergency situations.

5.2. Low-Level BC-PSSs Framework for Sustainable Furniture

Figure 11 shows a conceptual low-level BC-PSSs framework on the system level. The role of BC-PSSs in sustainable furniture is specifically introduced. During the design process, collaboration between different jobs, such as design briefs, CAD, and prototypes, is established, and a database of product design information (B1-1.2) is formed by recording personal work information (B1-1.1) to strengthen trust among professionals (B1-1). Creators upload creative content to the intellectual property protection system of BC to form an intellectual property database (B1-2.3) to protect the creator's intellectual rights (B1-2). The advantages of BC include that it supports more accurate planned inventory (B1-3) by managing supplier resources (B1-3.1), negotiated contracts (B1-3.2), orders and executions (B1-3.3), and others (B1-3.4). Smooth production is achieved (B1-4) by performing material forecasting (B1-4.1), production element lists (B1-4.2), standardized production (B1-4.3), and others (B1-4.4). During transportation, BC supports the development of reasonable operating plans (B2-1) to enable containers or products from loading to real-time delivery (B2-2) by tracking the status of product components (B2-2.1), real-time information of transport drivers (B2-2.2), control transportation plans (B2-2.3), and others (B2-2.4). In the process of product distribution, BC can trace the source of the product (B2-3.1), product packaging data (B2-3.2), product QR code (B2-3.3), and others (B2-3.4), which can prevent the false advertising and publicity of the product (B2-3); improve the transparency of the product, materials, and components (B2-4) (B2-4.1); and increase consumer trust. With the support of BC, it is possible to protect consumers' rights and interests in repair services (B3-1) by defining product maintenance standards (B3-1.1), establishing maintenance institutions and services (B3-1.2), among others (B3-1.3), and extending the lifecycle of products by creating smart contracts (B3-2.2) to ensure the validity of contracts (B3-2). BC improves product recovery rates (B3-3) by effectively analyzing the degree of damage to products (B3-3.1), product recovery value (B3-3.2), product condition assessment (B3-3.3), and others (B3-3.4). In addition, it can establish a reliable waste recycling system (B3-4), classify materials (B3-4.1), and rationally dispose of various materials (B3-4.2), among other actions (B3-4.3), to reduce environmental pollution. Throughout the lifecycle, BC ensures the secure storage and efficient transmission of all kinds of data in the design, procurement, production, transportation, distribution, maintenance, recycling, and disposal phases (B1-5, B2-5, B3-5).



Figure 11. Low-level BC-PSSs framework for sustainable furniture (generated by authors).

6. Review and Refinement of the Conceptual BC-PSSs Framework for Sustainable Furniture

6.1. Industry-Reviewed BC-PSSs Framework

The conceptual BC-PSSs framework was accepted and presented at the 2021 CSIAM-BTAF Conference (no official publication) and reviewed by industry experts and scholars in the form of pre-interview questionnaires and interviews [78]. The content includes a threedimensional review of the structure, content, and process of the high-level and low-level BC-PSSs conceptual frameworks, which are conducted in the form of participants first scoring the high-level and low-level conceptual BC-PSSs frameworks based on their personal views (scoring criteria of 1—strongly disagree, 2—disagree, 3—agree, and 4—strongly agree), and then presenting their own views and suggestions during the interview. The details are in Appendix A. The pre-interview questionnaire and interview were analyzed by SPSS and NVivo software, respectively. A total of seven industry experts and academics participated in this review, and Table 3 lists the demographic characteristics of the participants.

Characteristics	п	Proportion (%)
Gender		
Male	5	71.4%
Female	2	28.6%
Occupation		
Industry expert	5	71.4%
University Professor	2	28.6%
Number of years working on		
blockchain-related research		
Less than 5 years	1	14.3%
5–10 years	4	57.1%
10–20 years	2	28.6%
More than 20 years	0	0%

Table 3. Demographic characteristics of the participants (n = 7).

According to the results of the questionnaire before the interview, the reliability analysis of the questionnaire survey results was carried out using SPSS 22 software, and Cronbach's Alpha was greater than 0.6 and close to 1, indicating that the questionnaire had good reliability, as shown in Table 4. According to the scoring results, participants believed that the conceptual BC-PSSs framework had a clear structure, appropriate content, and clear process, as shown in Table 5.

Table 4. Reliability statistics (exported from SPSS 22).

Cronbach's Alpha	The Number of Projects
0.932	6

Table 5. Mean value of the conceptual high-level and low-level BC-PSSs framework (responses of industry experts and academics).

Evaluation Items	High-Level	Low-Level
Clarity of the structure	3.71	3.57
Appropriateness of content	3.43	3.43
Clarity of process	3.57	3.43

Based on the results of the interview, the original text of the interview was analyzed qualitatively using NVivo 11 software. Table 6 shows the core nodes, number of nodes, representative nodes, and representative interview texts. The interview transcript is coded into a total of five nodes, including "stakeholders", "strategies", "technologies", "institutions", and "policies". Participants in stakeholder-related issues mainly talked about correlation of "data", "data update", "collaborative communication", and "information sharing"; in the policy-related issues, the participants mainly mentioned "decentralized management", "mathematical model", "reward system", and "lifecycle management"; in the institution-related issues, the participants mainly involved "value assessment system", "a unified platform", and "the central institution"; in technology-related issues, the participants mainly mentioned "sues, the participants in policy-related issues mainly involved "corporate policy" and "the government's policy". Figure 12 shows the node distribution.

Core Nodes	Number of Nodes	Representative Nodes	Representative Interview Texts
Stakeholders	10	Correlation of data Data update Collaborative communication Information sharing	Blockchain can enhance data correlation between designers and consumers.
Strategies	8	Decentralized management Mathematical model Reward system Lifecycle management	Blockchain solves the problem by building relationships between different businesses in a decentralized setting.
Institutions	7	Value assessment system A unified platform The central institution	Blockchain enables data to be recorded on a chain so that its value can be assessed. This requires the enterprise to establish the value evaluation system, through some standards and rules to evaluate the value.
Technologies	7	Copyright protection Roots Wisdom contracts	Copyright protection is now a relatively mature technology in blockchain.
Policies	4	Corporate policy The government's policy	Whether blockchain can be successfully implemented in this industry depends on the willingness of enterprises to do it, and needs the support of enterprises.

 Table 6. The encoding results of the interview transcripts via NVivo 11 software.



Figure 12. Node distribution (exported from NVivo 11 software).

Subsequently, NVivo11 software was used to analyze the interview word frequency. As shown in Figure 13, the participants repeatedly mentioned the following words "value", "blockchain", "data", "design", "contracts", "need", "product", "consumer", and "contribution". Data transmission and storage for the sustainable furniture lifecycle and collaboration between stakeholders have attracted the attention of respondents. Introducing BC into a sustainable furniture lifecycle can improve trust among stakeholders, enhance data connectivity, and reduce information errors. In order to encourage the value contribution of stakeholders, a value evaluation system can be established to record value through pricing, appraisal, and performance. However, BC is implemented in the lifecycle stage of sustainable furniture, which requires central agency management and needs to standardize furniture product components to reduce the cost of BC establishment.



Figure 13. Interview word frequency (exported from NVivo 11 software).

During the interviews, participants noted that BC has great potential in enhancing PSSs IM and addressing challenges in the sustainable furniture lifecycle. In addition, participants made a number of useful suggestions to strengthen the content of the conceptual BC-PSSs framework, including the following:

- 1. BC can facilitate the design of incentives to encourage stakeholders to contribute personal knowledge.
- 2. BC can strengthen the data relevance between designers and consumers. There is an information asymmetry between the designer and the consumer through the enterprise or factory.
- 3. For the challenges of the furniture lifecycle, BC may not be able to prevent false advertising and publicity, just as we cannot prevent car accidents. Of course, BC can reduce false advertising and publicity to a certain extent because it can be traced, queried, and verified. This lets consumers understand the source of products, product materials, and production information.
- 4. BC enables data to be recorded on the chain for value assessment. This requires enterprises to establish a value evaluation system and evaluate value through some standards and rules.

6.2. The Updated BC-PSSs Framework

Refinements were made at the user layer and system layer of the high-level conceptual BC-PSSs IM framework as suggested by industry experts and scholars. As shown in Figure 14, at the user layer, "reward" was added in A1 and upgraded to "reward mechanism facilitates stakeholders communication and information sharing", In addition, we added A4 "stakeholders value recording and evaluation". At the system layer, we modified B2-3 "prevent false advertising and publicity" to "reduce false advertising and publicity".



Figure 14. The updated high-level BC-PSSs framework (generated by authors).

7. Discussion

7.1. Conceptual BC-PSSs Framework Contributes to Sustainable Furniture

In this paper, the conceptual framework of BC-PSSs is established to apply BC and PSSs IM to the whole lifecycle of furniture to promote its sustainable development. The framework presents eight stages of furniture design, procurement, manufacture, transportation, distribution, maintenance, recycle, and disposal, and it fully responds to the challenges from the user layer, system layer, and knowledge layer, respectively. In the user layer, BC and PSSs IM are used to address stakeholder-related issues such as collaboration, information sharing, valuation, and data storage. In the system layer, BC and PSSs IM are used to address stages of the lifecycle, such as proper transportation planning. In the knowledge layer, BC and PSSs IM are used to deal with issues related to knowledge management, such as product knowledge and material knowledge.

The BC-PSSs framework shows that PSSs IM provides effective product–service lifecycle activities for furniture lifecycle management, linking reliable products and value-added services as a system. Through the furniture product lifecycle knowledge management, we can reasonably and effectively provide required knowledge for stakeholders. At the same time, providing customers with maintenance and recycling services is conducive to forming a closed loop of materials and reducing the resource consumption of furniture products. In addition, it can contribute to promoting stakeholder collaboration and improving corporate competitiveness and resource productivity. BC can monitor products throughout the lifecycle of furniture and gather important information and data to better design, produce, sell, use, and recycle. BC also provides a way to enhance the feasibility of sharing data across the product lifecycle, facilitate stakeholder engagement, and bring trust to all parties. In addition, BC makes the furniture lifecycle transparent and traceable, which can further close the product lifecycle and reduce gas emissions and waste generation. BC can not only bring advantages to the furniture lifecycle, but also improve the traditional PSSs lifecycle management from multiple aspects, providing support for improving data storage transmission, security, trust, privacy, and other issues in the furniture PSSs lifecycle. BC can reasonably manage the products, services, information, and data generated throughout the lifecycle of PSSs IM, and securely track and trace products, helping PSSs IM effectively trigger services and delivery services and overcome the problem of data insecurity and low trust between stakeholders.

This study shows that BC has the potential to promote the implementation of PSSs IM in sustainable furniture and jointly address the challenges of sustainable furniture lifecycle, which is mainly reflected in the protection of creators' intellectual property rights, effective inventory management of raw material procurement, improvement of standardized specifications and production of product elements, effective execution of logistics planning and transportation management, improvement of consumer trust in the distribution process, improvement of maintenance services and contract effectiveness, improvement of waste recycling and treatment systems, and ensuring safe storage and efficient transmission of data.

7.2. The Potential of BC-PSSs to Quantify Design Value in Sustainable Furniture

According to Section 4.2.1 of this paper, the design casts a deep "environmental shadow" on the downstream processes of the lifecycle and has an impact on the entire lifecycle. Therefore, the decisions made during the design phase are crucial and come mainly from the designer. According to Section 4.2.5, the BC technology timestamp, asymmetric encryption, tamper-proof features, and transparency can bring advantages to protect the rights and interests of designers. Therefore, this section specifically discusses how BC-enhanced PSSs IM encourages designers to create more design value and discusses its potential in value quantification.

Currently, design is increasingly present in the context of companies, and the value created by its implementation, as well as the possibilities of management, become crucial. Design can innovate the image of a business, optimize costs, and develop innovative products and services; help maintain product quality in terms of function, production, economic, and social culture; and improve the competitiveness of enterprises [82]. The stakeholders (designers) involved in the design activities play an important role at this stage, and the creative freedom of designers can affect the materials and characteristics of the final product [1]. Designers can choose and determine the material, formal and practical, and some other aspects of furniture during the design process, influencing the emotional response of consumption and thus consumer behavior [125]. For example, designers can raise consumers' awareness of the environment by designing websites, advertisements, brochures, and other promotional materials, and they can communicate with consumers about the benefits of using such products [109]. Therefore, the value of design activities is not only reflected in commercial value and material benefits, but also in ecological values and social responsibility.

The value created by designers needs to be recognized and acknowledged because there is a positive cycle between value creation and member benefits; that is, value creation requires the participation and contribution of members, and members need to be able to enjoy the benefits of value creation. However, furniture designers currently lack the encouragement to express their ideas and designs. Each stakeholder in an organization has a certain amount of personal knowledge, and due to the lack of reasonable, feasible, and mandatory sharing reward mechanisms, contributors are not motivated to share knowledge and are reluctant to contribute their knowledge to other organizations [115,126,127]. This will affect designers to create more valuable products to some extent.

The "blockchain revolution" has brought a paradigm shift to the creative economy [128]. First of all, BC technology can store and manage information, such as intellectual property rights, and reflect the design value of creators through secure sales and transactions. BC makes the lifecycle management of intellectual property works more convenient and transparent, including creation, registration, and transaction [103]. Through trading, creators can obtain the economic benefits of their works more directly and realize the value of intellectual property works. At the same time, this can stimulate the enthusiasm and motivation of creators [129], thereby inspiring people to make meaningful contributions to a common goal. Second, BC is a means of recording specific states of affairs agreed upon by the network in a secure and verifiable manner [130]. Data on the BC are time-stamped and immutable, which is ideal for tracking members' activities and ensuring accountability for all data published [131]. For example, after the product design is completed, the BC can record the raw materials and energy consumed by the product, the exhaust gas and waste emitted, the environmental impact caused by use, the degree of product material recycling, the energy consumption and pollution generated by waste disposal, etc., so as to evaluate the social and ecological value created by the designer. In addition, one of the most promising features of BC technology as a medium for value recording is decentralized value recording, capable of encapsulating qualitatively different contributions [132]. Now, more and more consensus mechanisms are being proposed, such as Proof of Value (PoV); PoV protocols provide a mechanism for decentralized consensus that determines the value of each contribution, which can equally reward smart contracts that generate value [133]. BC is used to record all design attempts in architectural design, recording the value contribution of creators in solving design problems [134].

Recently, the concept of non-fungible tokens (NFTs) has emerged in the BC world as a unique way to attribute ownership to digital assets [135]. NFTs are tradable rights to digital assets (pictures, music, movies, and virtual creations) where ownership is recorded in a BC smart contract. NFTs have been used as a way to reward content creators, provide intellectual property protection for creators, make it easier to buy and sell patents, and enable creators to benefit from innovation [136]. Due to the rapid development of various new technologies, the digital economy will enter the 3.0 era, and BC and the metaverse are the infrastructure that will support the digital economy 3.0. Digital assets will become a very important way to display value in the metaverse [137]. Paired with the metaverse, NFTs represent a major advancement and revolution in the field of virtual reality and BC, providing content creators with new ways to express their unique and valuable work [138].

7.3. The Potential of BC-PSSs to Improve Consumer Behavior in Sustainable Furniture

According to Sections 2.1 and 4.2.1 of this article, the sustainability of furniture is largely influenced by consumer behavior, especially in terms of product longevity. Therefore, improving consumer behavior is particularly important for environmental protection and sustainable development. At the same time, according to Section 4.2.5, BC technology can bring advantages to consumers through traceability, irreversibility, and smart contracts. Therefore, this section will discuss specifically how BC-enhanced PSSs IM brings potential to promote sustainable consumer behavior.

Consumer behavior is an act or activity of a consumer to select, use, and dispose of a product or service. This includes not only purchases, but also ideas, suggestions, and actions related to consumer purchasing behavior. The essence of consumer behavior is the activity of consumers to meet their needs by purchasing products or services [139]. In the consumer behavior literature, environmentally sustainable behavior is often described as green consumption activities [140,141]. Consumer green behavior is an important sub-area of sustainable development, including recycling, waste reduction, local consumption, energy saving, reuse, maintenance, and other ecological behaviors, which are often summarized

into three main categories of behavior, purchasing behavior, use behavior, and end-of-use behavior [142].

The general transformation of consumer behavior is the core of sustainable consumption [143]. Researchers have proposed a variety of ways to change consumer behavior, for example, through the use of mandatory regulatory and legal mechanisms to promote waste treatment services [144]; providing adequate information through educational services has been shown to be effective in changing consumer behavior, including educating consumers about the proper installation, use, maintenance, and disposal of products [145]. With the enhancement of consumers' awareness of environmental protection, the furniture industry has introduced the concept of green furniture and conveyed the characteristics of environmental protection and social expectations of products to the final consumers through the form of eco-labels [146]. However, due to various counterfeiting problems, consumers have doubts and distrust of eco-labeled products [147,148]. Therefore, it hinders the promotion of eco-friendly furniture. BC technology can improve information traceability and increase consumer confidence in their products, providing consumers with true and transparent information. In addition, BC records help monitor and prove the environmental sustainability of the products produced to customers [13]. IKEA has applied BC technology to green and sustainable supply chain management as a guarantee that the entire process is controlled and that the final product is indeed made of said specific wood [149]. BC can facilitate the design of incentives to encourage green behavior among consumers, improve visibility, increase efficiency, and support performance monitoring and reporting. At the same time, native tokens on digital platforms can promote green behavior for consumers and other participants during the product lifecycle [142]. Mechanisms, such as BC certification of suppliers, inspectors, and IoT sensors, can be used to verify the ecological behavior of individuals [150], and consumers' sustainable behavior can be rewarded through tokenization. Some organizations are already using BC to encourage sustainable behavior among consumers, such as CarbonX rewarding consumers who purchase carbon-neutral products based on assessing the carbon footprint of products and services, and Recycle To Coin rewarding consumers who return recyclables to collection sites [142].

In summary, we can enhance product lifecycle visibility by incentivizing individual participation in promoting green behaviors, facilitating the tracking of a product's actual footprint, and improving the efficiency of emissions trading schemes by reducing fraud and improving system fidelity [149]. BC plays a particularly important role in sustainability, with its potential to alleviate some of the sustainability-related challenges and facilitate the realization of a CE [151].

8. Conclusions

With the increasingly serious problems of resource shortage and environmental damage, especially under the impact of COVID-19, sustainable development has become even more urgent. As an indispensable item in people's daily life, furniture has the characteristics of consuming a large amount of raw materials and generating waste, and it still needs to continue to explore and make progress in the development of sustainability. The PSSs can effectively manage the lifecycle and has the potential to develop sustainable furniture and promote CE, so this paper applies the PSSs to the lifecycle of furniture, carries out lifecycle knowledge management, promotes stakeholder collaboration, forms a material closed loop in the lifecycle, improves resource productivity, and enhances enterprise competitiveness. Although the PSSs brings a series of benefits to the sustainable development of the furniture industry, it is still necessary to consider how to ensure the secure storage and transmission of information and data in the PSSs lifecycle, as well as the rights and interests of stakeholders and other problems. BC has the characteristics of distributed database, peer-to-peer transactions, transaction automation, anonymity, and irreversibility of records in terms of ensuring data security, opening up a treasure trove of renewable digital possibilities for the goal of sustainable CE. This paper uses the potential advantages

of BC to solve the challenges of sustainable furniture PSSs from the perspective of lifecycle and further promote the sustainable development of the furniture industry.

This paper attempts for the first time to construct an overview BC-enhanced PSSs framework for sustainable furniture from the whole lifecycle stage; it constructs a highlevel framework including three levels of user, system, and knowledge from a strategic perspective, and it constructs a low-level framework of the system layer according to specific details. The framework adopts a mixed quantitative and qualitative method, and first quantitatively analyzes the potential relationship between furniture, PSSs, and BC through VOSviewer. Then, it qualitatively analyzes the lifecycle stages of sustainable furniture, the advantages of PSSs in promoting sustainable furniture, and the potential of BC to enhance PSSs IM to further promote sustainable furniture and solve related challenges. Subsequently, the conceptual framework of BC-PSSs is constructed, reviewed, and refined through pre-interview questionnaires and follow-up interviews with industry experts and scholars. In addition, this article explores the potential of BC-PSSs to quantify the value of designers at the design stage through intellectual property, reward mechanisms, and NFTs to promote the value contribution of designers. BC-PSSs has the potential to improve consumer purchase, use, and post-use behavior to promote sustainable consumer behavior in furniture consumption. The study shows the potential promise of using a PSSs in promoting sustainability and CE in the furniture industry, as well as the potential of BC to enhance PSSs IM to solve furniture lifecycle challenges. The results can provide reference for researchers and policy makers in related industries. More importantly, the study contributes to the achievement of SDGs 11 (Sustainable Cities and Communities) and SDGs 12 (Responsible Consumption and Production). However, the quantitative analysis based on VOSviewer software in this paper mainly focuses on the WOS core collection database, and subsequent studies can quantitatively analyze multiple databases, such as Scopus and Science Direct. At the same time, the conceptual BC-PSSs framework has been reviewed by industry experts and scholars, and more work needs to be done in the future to ensure its practical feasibility and practical significance.

Author Contributions: Conceptualization, Z.L., J.L. and Q.Y.; methodology, Z.L., J.L., Q.Y., M.O. and P.D.; software, J.L.; validation, Z.L., J.L., Q.Y., M.O. and P.D.; formal analysis, Z.L. and J.L.; investigation, Z.L., J.L. and Q.Y.; resources, Z.L. and Q.Y.; data curation, Z.L. and J.L.; writing—original draft preparation, Z.L., J.L. and Q.Y.; writing—review and editing, Z.L., J.L., Q.Y., M.O. and P.D.; visualization, J.L.; supervision, Z.L.; project administration, Z.L. and Q.Y.; funding acquisition, Z.L. and Q.Y. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by Department of Education of Guangdong Province, grant number j2jw-C9223037, and Guangdong Provincial Department of Science and Technology 2022 Overseas Famous Teacher Project: "Behavior and Service Design Course for Sustainable Youth Development City Construction (SYCBD)".

Data Availability Statement: The data presented in this study are available on request from the corresponding author. The data are not publicly available due to privacy.

Acknowledgments: The authors would like to thank all the people who support this research including three reviewers for their valuable advice to make the paper more constructive. Z.L. would like to thank Hitomi for her song of LAST EXILE for bringing him the touch of his youth and encouraging him to go forward, because of love, the same with today's covid-19 leading to self-exile when final proofreading, which is similar situation as the blockchain to product service systems. J.L. would like to gratitude the South China University of Technology for its rich learning resources, and her supervisor for his guidance and encouragement.

Conflicts of Interest: The authors declare no conflict of interest.

32 of 37

4

4

4

Appendix A

The purpose of this survey is to review and refine the conceptual BC-PSSs framework for sustainable furniture. The content includes a three-dimensional review of the structure, content, and process of the high-level and low-level BC-PSSs conceptual frameworks, which are conducted in the form of participants first scoring the high-level and lowlevel conceptual BC-PSSs frameworks based on your personal views (scoring criteria of 1—strongly disagree, 2—disagree, 3—agree, and 4—strongly agree), and then presenting your own views and suggestions.

- 1. High-level BC-PSSs framework for sustainable furniture
- (1) The structure of the high-level framework is clear. 1 2 3 4
- (2) The content of the high-level framework is appropriate. 1 2 3 4
- (3) The process of the high-level framework is clear. 1 2 3 4

2. Low-level BC-PSSs framework for sustainable furniture

- (1) The structure of the low-level framework is clear. 1 2 3 4
- (2) The content of the low-level framework is appropriate. 1 2 3 4
- (3) The process of the low-level framework is clear. 1 2 3 4
- 3. The BC potential to overcome challenges for PSSs IM in sustainable furniture
- (1) Protection of the intellectual property rights of creators. 1 2 3 4
- (2) Effective inventory management of raw material procurement. 1 2 3
- (3) Improving the standardization and production of product elements. 1 2
 3 4
- (4) Effectively implement logistics planning and transportation management. 1 2
 3 4
- (5) Improving consumer trust in the distribution process. 1 2 3 4
- (6) Improving maintenance services and contract effectiveness. 1 2 3
- (7) Improving the waste recycling and treatment system. 1 2 3 4
- (8) Ensuring secure storage and efficient transmission of data. 1 2 3
- 4. Further Comments Please add any additional comments about the conceptual BC-PSSs framework in the space below. Thank you very much for participating in this study!

References

- 1. Barbaritano, M.; Bravi, L.; Savelli, E. Sustainability and Quality Management in the Italian Luxury Furniture Sector: A Circular Economy Perspective. *Sustainability* **2019**, *11*, 3089. [CrossRef]
- Stahel, W.R. Policy for material efficiency-sustainable taxation as a departure from the throwaway society. *Philos. Trans. R. Soc. A* 2013, 371, 20110567. [CrossRef] [PubMed]
- 3. Kristensen, H.S.; Remmen, A. A framework for sustainable value propositions in product-service systems. *J. Clean. Prod.* 2019, 223, 25–35. [CrossRef]
- 4. Mont, O.K. Clarifying the concept of product-service system. J. Clean. Prod. 2002, 10, 237–245. [CrossRef]
- Pieroni, M.P.P.; McAloone, T.C.; Pigosso, D.C.A. Configuring New Business Models for Circular Economy through Product-Service Systems. Sustainability 2019, 11, 3732. [CrossRef]
- 6. Johnson, E.; Plepys, A. Product-Service Systems and Sustainability: Analysing the Environmental Impacts of Rental Clothing. *Sustainability* **2021**, *13*, 2118. [CrossRef]
- 7. Li, C.Z.; Chen, Z.; Xue, F.; Kong, X.T.R.; Xiao, B.; Lai, X.; Zhao, Y. A blockchain- and IoT-based smart product-service system for the sustainability of prefabricated housing construction. *J. Clean. Prod.* **2021**, *286*, 125391. [CrossRef]
- 8. Hankammer, S.; Steiner, F. Leveraging the sustainability potential of mass customization through product service systems in the consumer electronics industry. *Procedia CIRP* **2015**, *30*, 504–509. [CrossRef]
- 9. Rymaszewska, A.; Helo, P.; Gunasekaran, A. IoT powered servitization of manufacturing—An exploratory case study. *Int. J. Prod. Econ.* **2017**, *192*, 92–105. [CrossRef]
- 10. Jing, Q.; Vasilakos, A.V.; Wan, J.; Lu, J.; Qiu, D. Security of the Internet of Things: Perspectives and challenges. *Wirel. Netw.* **2014**, 20, 2481–2501. [CrossRef]
- Niu, X.; Wang, M.; Qin, S. Product design lifecycle information model (PDLIM). Int. J. Adv. Manuf. Technol. 2022, 118, 2311–2337. [CrossRef]
- 12. Wang, R.; Zhong, D.; Zhang, Y.; Yu, J.; Li, M. A multidimensional information model for managing construction information. *J. Ind. Manag. Optim.* **2015**, *11*, 1285–1300. [CrossRef]

- 13. Upadhyay, A.; Mukhuty, S.; Kumar, V.; Kazancoglu, Y. Blockchain technology and the circular economy: Implications for sustainability and social responsibility. *J. Clean. Prod.* 2021, 293, 126130. [CrossRef]
- Watanabe, K.; Okuma, T.; Takenaka, T. Evolutionary design framework for Smart PSS: Service engineering approach. Adv. Eng. Inform. 2020, 45, 101119. [CrossRef]
- Despeisse, M.; Baumers, M.; Brown, P.; Charnley, F.; Ford, S.J.; Garmulewicz, A.; Knowles, S.; Minshall, T.; Mortara, L.; Reed-Tsochas, F.P. Unlocking value for a circular economy through 3D printing: A research agenda. *Technol. Forecast. Soc. Chang.* 2017, 115, 75–84. [CrossRef]
- 16. Lopes De Sousa Jabbour, A.B.; Chiappetta Jabbour, C.J.; Godinho Filho, M.; Roubaud, D. Industry 4.0 and the circular economy: A proposed research agenda and original roadmap for sustainable operations. *Ann. Oper. Res.* **2018**, *270*, 273–286. [CrossRef]
- 17. Tapscott, D.; Tapscott, A. Blockchain Revolution: How the Technology Behind Bitcoin is Changing Money, Business, and the World; Penguin: London, UK, 2016.
- 18. Underwood, S. Blockchain beyond bitcoin. Commun. ACM 2016, 59, 15–17. [CrossRef]
- 19. Lakhani, K.R.; Iansiti, M. The truth about blockchain. *Harv. Bus. Rev.* 2017, 95, 119–127.
- Sawant, T.; Idayakumar, P.; Sabkale, A.; Pampattiwar, K. Decentralized EHR Storage Using Blockchain. ECS Trans. 2022, 107, 6397–6405. [CrossRef]
- Salau, O.; Adeshina, S.A. Secure Document Verification System Using Blockchain. In Proceedings of the 2021 1st International Conference on Multidisciplinary Engineering and Applied Science (ICMEAS), Abuja, Nigeria, 15–16 July 2021; p. 7.
- Cheng, S.; Mu, S.; He, L.; Huang, J.; Zhao, L.; Wang, Q.; Zhang, P. A solution to electronic license based on Blockchain. In Proceedings of the 2020 International Conference on Computer Engineering and Application (ICCEA), Guangzhou, China, 18–20 March 2020; pp. 6–10.
- 23. Kim, Y.S.; Lee, H.J. Product-Service Systems Design Approach: Servitization of a Small Furniture Manufacturing Company. *Korean Soc. Sci. Art* **2015**, *20*, 159–171. [CrossRef]
- 24. Ohgren, M.; Milios, L.; Dalhammar, C.; Lindahl, M. Public procurement of reconditioned furniture and the potential transition to product service systems solutions. *Procedia CIRP* **2019**, *83*, 151–156. [CrossRef]
- 25. Costa, F.; Prendeville, S.; Beverley, K.; Teso, G.; Brooker, C. Sustainable product-service systems for an office furniture manufacturer: How insights from a pilot study can inform PSS design. *Procedia CIRP* **2015**, *30*, 66–71. [CrossRef]
- Baygin, N.; Baygin, M.; Karakose, M. Blockchain Application in Mass Customization: A Furniture Sector Example. In Proceedings of the 2020 International Conference on Data Analytics for Business and Industry: Way towards a Sustainable Economy (ICDABI), Sakheer, Bahrain, 26–27 October 2020; p. 5.
- 27. Jaeger, B.; Bach, T.; Pedersen, S.A. A Blockchain Application Supporting the Manufacturing Value Chain. In *IFIP International Conference on Advances in Production Management Systems*; Springer: Cham, Switzerland, 2019; pp. 466–473. [CrossRef]
- Sheng, S.W.; Wicha, S. The Proposed of a Smart Traceability System for Teak Supply Chain Based on Blockchain Technology. In Proceedings of the 2021 Joint International Conference on Digital Arts, Media and Technology with ECTI Northern Section Conference on Electrical, Electronics, Computer and Telecommunication Engineering, Cha-am, Thailand, 3–6 March 2021; pp. 59–64.
- 29. Annarelli, A.; Battistella, C.; Nonino, F. Product service system: A conceptual framework from a systematic review. *J. Clean. Prod.* **2016**, 139, 1011–1032. [CrossRef]
- Cordella, M.; Hidalgo, C. Analysis of key environmental areas in the design and labelling of furniture products: Application of a screening approach based on a literature review of LCA studies. *Sustain. Prod. Consum.* 2016, *8*, 64–77. [CrossRef]
- 31. Zion Market Research Report. "Furniture Market by Type (Tables, Chairs, Beds, Sofas, Cupboards, and Others), By Material (Wood, Plastic, Metal, and Others (Glass, Leather, and Rubber)), and By Application (Residential and Commercial): Global Industry Perspective, Comprehensive Analysis and Forecast, 2017–2024". Available online: https://www.globenewswire.com/news-release/2018/09/07/1567975/0/en/Global-FurnitureMarket-Will-Reach-USD-472-30-Billion-By-2024-Zion-Market-Research.html (accessed on 13 May 2019).
- Gonzalez-Garcia, S.; Gasol, C.M.; Garcia Lozano, R.; Teresa Moreira, M.; Gabarrell, X.; Rieradevall I Pons, J.; Feijoo, G. Assessing the global warming potential of wooden products from the furniture sector to improve their ecodesign. *Sci. Total Environ.* 2011, 410, 16–25. [CrossRef] [PubMed]
- Cavagnaro, E.; Altena, D.; Seidel, S. Saving the commons: Exploring the potential contribution of the furniture industry. In Proceedings of the Twenty-First Annual Meeting of the International Association for Business and Society, 2010, Banff, Alberta, Canada, 25 March 2010; Volume 21, pp. 32–38.
- 34. Besch, K. Product-service systems for office furniture: Barriers and opportunities on the European market. *J. Clean. Prod.* 2005, *13*, 1083–1094. [CrossRef]
- Shoji, Y.; Nakao, N.; Ueda, Y.; Kakizawa, H.; Hirai, T. Preferences for certified forest products in Japan: A case study on interior materials. For. Policy Econ. 2014, 43, 1–9. [CrossRef]
- 36. Bobba, S.; Ardente, F.; Mathieux, F. Environmental and economic assessment of durability of energy-using products: Method and application to a case-study vacuum cleaner. *J. Clean. Prod.* **2016**, 137, 762–776. [CrossRef]
- Lihra, T.; Graf, R. Multi-channel communication and consumer choice in the household furniture buying process. *Direct Mark. Int. J.* 2007, 1, 146–160. [CrossRef]

- Forrest, A.; Hilton, M.; Ballinger, A.; Whittaker, D. Circular Economy Opportunities in the Furniture Sector; Eunomia Research & Consulting Ltd for European Environmental Bureau: Bussels, Belgium, 2017.
- 39. Vollmer, A. Büromöbelleasing-Chancen eines öko-effizienten Dienstleistungskonzeptes: Theoretische Grundlagen und erste praktische Ansätze; Diplomarbeiten Agentur diplom.de: Hamburg, Germany, 1999.
- Korhonen, J.; Nuur, C.; Feldmann, A.; Birkie, S.E. Circular economy as an essentially contested concept. J. Clean. Prod. 2018, 175, 544–552. [CrossRef]
- The Unite Nation. Available online: https://www.un.org/sustainabledevelopment/sustainable-development-goals/ (accessed on 29 September 2022).
- MacArthur, E. Towards the Circular Economy, Economic and Business Rationale for an Accelerated Transition; Ellen MacArthur Foundation: Cowes, UK, 2013; pp. 21–34.
- Beuren, F.H.; Ferreira, M.G.G.; Miguel, P.A.C. Product-service systems: A literature review on integrated products and services. J. Clean. Prod. 2013, 47, 222–231. [CrossRef]
- 44. Lewandowski, M. Designing the Business Models for Circular Economy-Towards the Conceptual Framework. *Sustainability* **2016**, *8*, 43. [CrossRef]
- 45. Lieder, M.; Rashid, A. Towards circular economy implementation: A comprehensive review in context of manufacturing industry. *J. Clean. Prod.* **2016**, *115*, 36–51. [CrossRef]
- 46. Goedkoop, M.J.; van Halen, C.; Te Riele, H.; Rommens, P. Product service systems, ecological and economic basics report Dutch Ministries of Environment (VROM) and Economic Affairs (EZ), 1999. Economic Affairs 1999. Available online: https: //docplayer.net/334668-Product-service-systems-ecological-and-economic-basics.html (accessed on 18 December 2022).
- 47. Tukker, A. Eight types of product–service system: Eight ways to sustainability? Experiences from SusProNet. *Bus. Strategy Environ.* **2004**, *13*, 246–260. [CrossRef]
- Cook, M.B.; Bhamra, T.A.; Lemon, M. The transfer and application of Product Service Systems: From academia to UK manufacturing firms. J. Clean. Prod. 2006, 14, 1455–1465. [CrossRef]
- 49. Manzini, E.; Vezzoli, C. A strategic design approach to develop sustainable product service systems: Examples taken from the 'environmentally friendly innovation'Italian prize. *J. Clean. Prod.* **2003**, *11*, 851–857. [CrossRef]
- 50. Komoto, H.; Tomiyama, T. Integration of a service CAD and a life cycle simulator. CIRP Ann. 2008, 57, 9–12. [CrossRef]
- 51. Hajimohammadi, A.; Cavalcante, J.; Gzara, L. Ontology for the PSS lifecycle management. Procedia CIRP 2017, 64, 151–156. [CrossRef]
- Cavalcante, J.; Gzara, L. Product-Service Systems lifecycle models: Literature review and new proposition. *Procedia CIRP 2018* 2018, 73, 32–38. [CrossRef]
- 53. Stoll, O.; Zou, W.; Rodel, E.; West, S. Value Co-creation in the Context of Digitally-Enabled Product-Service Systems. In *IFIP* Advances in Information and Communication Technology; Springer: Cham, Switzerland, 2021; Volume 629, pp. 337–344. [CrossRef]
- 54. Yoo, M.; Grozel, C.; Kiritsis, D. Closed-Loop Lifecycle Management of Service and Product in the Internet of Things: Semantic Framework for Knowledge Integration. *Sensors* **2016**, *16*, 1053. [CrossRef]
- 55. Wang, P.P.; Ming, X.G.; Li, D.; Kong, F.B.; Wang, L.; Wu, Z.Y. Status review and research strategies on product-service systems. *Int. J. Prod. Res.* **2011**, *49*, 6863–6883. [CrossRef]
- Schuh, G.; Gudergan, G.; Feige, B.A.; Buschmeyer, A.; Krechting, D. Business Transformation in the manufacturing industry— How information acquisition, analysis, usage and distribution affects the success of Lifecycle-Product-Service-Systems. *Procedia CIRP* 2015, 30, 335–340. [CrossRef]
- 57. van der Laan, A.Z.; Aurisicchio, M. A framework to use product-service systems as plans to produce closed-loop resource flows. *J. Clean. Prod.* **2020**, 252. [CrossRef]
- Reim, W.; Parida, V.; Örtqvist, D. Product–Service Systems (PSS) business models and tactics—A systematic literature review. J. Clean. Prod. 2015, 97, 61–75. [CrossRef]
- 59. Tukker, A. Product services for a resource-efficient and circular economy—A review. J. Clean. Prod. 2015, 97, 76–91. [CrossRef]
- Xia, M.; Xie, Z.; Lin, H.; He, X. Synergistic Mechanism of the High-Quality Development of the Urban Digital Economy from Blockchain Adoption Perspective-A Configuration Approach. J. Theor. Appl. Electron. Commer. Res. 2022, 17, 704–721. [CrossRef]
- 61. Alshboul, O.; Shehadeh, A.; Almasabha, G.; Almuflih, A.S. Extreme Gradient Boosting-Based Machine Learning Approach for Green Building Cost Prediction. *Sustainability* **2022**, *14*, 6651. [CrossRef]
- 62. Amuso, V.; Poletti, G.; Montibello, D. The digital economy: Opportunities and challenges. Glob. Policy 2019, 11, 124–127. [CrossRef]
- 63. Calvão, F.; Archer, M. Digital extraction: Blockchain traceability in mineral supply chains. Political Geogr. 2021, 87, 102381. [CrossRef]
- 64. Seebacher, S.; Schüritz, R. Blockchain technology as an enabler of service systems: A structured literature review. In *Exploring Services Science (IESS 2017)*; Springer: Cham, Switzerland, 2017; pp. 12–23.
- 65. Huang, J.; Li, S.; Thurer, M. On the Use of Blockchain in Industrial Product Service Systems: A critical Review and Analysis. *Procedia CIRP* **2019**, *83*, 552–556. [CrossRef]
- 66. Moll, J.; Yigitbasioglu, O. The role of internet-related technologies in shaping the work of accountants: New directions for accounting research. *Br. Account. Rev.* 2019, *51*, 100833. [CrossRef]
- 67. Mentsiev, A.U.; Guzueva, E.R.; Yunaeva, S.M.; Engel, M.V.; Abubakarov, M.V. Blockchain as a technology for the transition to a new digital economy. *J. Phys. Conf. Ser.* **2019**, *1399*, 033113. [CrossRef]
- 68. Viriyasitavat, W.; Xu, L.D.; Bi, Z.; Pungpapong, V. Blockchain and Internet of Things for Modern Business Process in Digital Economy-the State of the Art. *IEEE Trans. Comput. Soc. Syst.* **2019**, *6*, 1420–1432. [CrossRef]

- 69. Boucher, P. How Blockchain Technology Could Change Our Lives: In-depth Analysis; European Parliament: Brussels, Belgium, 2017.
- 70. Szabo, N. Smart Contracts. Available online: http://www.fon.hum.uva.nl/rob/Courses/InformationInSpeech/CDROM/ Literature/LOTwinterschool2006/szabo.best.vwh.net/smart.contracts.html (accessed on 8 February 2018).
- 71. Sakho, S.; Zhang, J.; Essaf, F.; Kiki, M.J.M. Blockchain: Perspectives and issues. J. Intell. Fuzzy Syst. 2019, 37, 8029–8052. [CrossRef]
- Hamida, E.B.; Brousmiche, K.L.; Levard, H.; Thea, E. Blockchain for enterprise: Overview, opportunities and challenges. In Proceedings of the Thirteenth International Conference on Wireless and Mobile Communications (ICWMC 2017), Nice, France, 23–27 July 2017.
- 73. Ghisellini, P.; Cialani, C.; Ulgiati, S. A review on circular economy: The expected transition to a balanced interplay of environmental and economic systems. *J. Clean. Prod.* **2016**, *114*, 11–32. [CrossRef]
- 74. Swan, M. Blockchain: Blueprint for a New Economy; O'Reilly Media: Newton, MA, USA, 2015.
- 75. Xu, M.; Chen, X.; Kou, G. A systematic review of blockchain. Financ. Innov. 2019, 5, 27. [CrossRef]
- 76. Van Eck, N.; Waltman, L. Software survey: VOSviewer, a computer program for bibliometric mapping. *Scientometrics* **2010**, *84*, 523–538. [CrossRef]
- 77. Li, H.; An, H.; Wang, Y.; Huang, J.; Gao, X. Evolutionary features of academic articles co-keyword network and keywords co-occurrence network: Based on two-mode affiliation network. *Physica A* **2016**, *450*, 657–669. [CrossRef]
- CSIAM Blockchain Committee. "Chinese Society of Industrial and Applied Mathematics BC Technology and Application Summit Forum 2021" (CSIAM-BTAF 2021). Available online: https://mp.weixin.qq.com/s/A0N1AqBKykt-UZM_hmLpgg (accessed on 23 September 2022).
- Forero, R.; Nahidi, S.; De Costa, J.; Mohsin, M.; Fitzgerald, G.; Gibson, N.; McCarthy, S.; Aboagye-Sarfo, P. Application of four-dimension criteria to assess rigour of qualitative research in emergency medicine. *BMC Health Serv. Res.* 2018, 18, 120. [CrossRef]
- Lincoln, Y.S.; Guba, E.G. But is it rigorous? Trustworthiness and authenticity in naturalistic evaluation. *New Dir. Program Eval.* 1986, 1986, 73–84. [CrossRef]
- 81. Purwaningsih, R.; Susanto, N.; Adiaksa, D.A.; Putri, A.A.A. Analysis of the eco-efficiency level in the dining table production process using life cycle assessment method to increase industry sustainability. *IOP Conf. Ser. Mater. Sci. Eng.* **2021**, 1072, 12014–12017. [CrossRef]
- 82. Bianco, I.; Thiebat, F.; Carbonaro, C.; Pagliolico, S.; Blengini, G.A.; Comino, E. Life Cycle Assessment (LCA)-based tools for the eco-design of wooden furniture. *J. Clean. Prod.* **2021**, *324*, 129249. [CrossRef]
- Chen, M.; Lyu, J.; Li, S.; Wu, X. Construction and implementation of a panel furniture design evaluation system at the design stage. *Adv. Mech. Eng.* 2017, *9*, 1687814017693945. [CrossRef]
- Mirabella, N.; Castellani, V.; Sala, S. LCA for assessing environmental benefit of eco-design strategies and forest wood short supply chain: A furniture case study. *Int. J. Life Cycle Assess.* 2014, 19, 1536–1550. [CrossRef]
- 85. Tsoulfas, G.T.; Pappis, C.P. Environmental principles applicable to supply chains design and operation. *J. Clean. Prod.* **2006**, *14*, 1593–1602. [CrossRef]
- 86. Rashid, A.; Asif, F.M.A.; Krajnik, P.; Nicolescu, C.M. Resource Conservative Manufacturing: An essential change in business and technology paradigm for sustainable manufacturing. *J. Clean. Prod.* **2013**, *57*, 166–177. [CrossRef]
- 87. Jeswiet, J.; Hauschild, M. EcoDesign and future environmental impacts. Mater. Des. 2005, 26, 629–634. [CrossRef]
- 88. Ramani, K.; Ramanujan, D.; Bernstein, W.Z.; Zhao, F.; Sutherland, J.; Handwerker, C.; Choi, J.; Kim, H.; Thurston, D. Integrated sustainable life cycle design: A review. *J. Mech. Des.* **2010**, *132*, 091004. [CrossRef]
- Sutopo, W.; Devi, A.; Hisjam, M.; Yuniaristanto, S. A model for procurement and inventory planning for export-oriented furniture industry in Indonesia: A case study. In Proceedings of the International Multiconference of Engineers and Computer Scientists, Hong Kong, China, 14–16 March 2012; pp. 1214–1217.
- González-García, S.; Lozano, R.G.; Moreira, M.T.; Gabarrell, X.; I Pons, J.R.; Feijoo, G.; Murphy, R.J. Eco-innovation of a wooden childhood furniture set: An example of environmental solutions in the wood sector. *Sci. Total Environ.* 2012, 426, 318–326. [CrossRef]
- 91. IHOBE. Guías Sectoriales de Ecodiseño Mobiliario. Available online: http://www.ihobe.net/Publicaciones/ficha.aspx?IdMenu= 750e07f4-11a4-40da-840c-0590b91bc032&Cod=03ded2c8-31b3-4eff-9df1-9b424a30b508&Tipo= (accessed on 23 July 2014).
- Mont, O. Introducing and Developing a Product-Service System (PSS) Concept in Sweden; The International Institute for Industrial Environmental Economics: Lund, Sweden, 2001; Available online: https://portal.research.lu.se/en/publications/introducingand-developing-a-product-service-system-pss-concept-i (accessed on 18 December 2022).
- 93. Manzini, E.; Vezzoli, C.A. *Product-Service Systems and Sustainability: Opportunities for Sustainable Solutions*; UNEP-United Nations Environment Programme: Nairobi, Kenya, 2002.
- 94. Borrello, M.; Caracciolo, F.; Lombardi, A.; Pascucci, S.; Cembalo, L. Consumers' perspective on circular economy strategy for reducing food waste. *Sustainability* **2017**, *9*, 141. [CrossRef]
- 95. Aurich, J.C.; Schweitzer, E.; Fuchs, C. Life Cycle Management of Industrial Product-Service Systems. In *Advances in Life Cycle Engineering for Sustainable Manufacturing Businesses*; Springer: London, UK, 2007; Volume 171, pp. 171–176. [CrossRef]
- 96. Chen, C.; Lan, K. Strategic Study of Knowledge Management Which Led into Furniture Design Industry—Taking Example by Taiwan Furniture Industry. In *International Conference on Human Interface and the Management of Information;* Springer: Berlin/Heidelberg, Germany, 2013; pp. 433–442.

- Zhou, C.; Song, W.; Liu, L.; Niu, Z. Blockchain Technology-Enabled Smart Product-Service System Lifecycle Management: A Conceptual Framework. In Proceedings of the 2020 IEEE 16th International Conference on Automation Science and Engineering (CASE), Hong Kong, China, 20–21 August 2020; pp. 1175–1180.
- Schuler, A.; Buehlmann, U. Identifying Future Competitive Business Strategies for the US Furniture Industry: Benchmarking and Paradigm Shifts; US Department of Agriculture, Forest Service, Northeastern Research Station: Newtown Square, PA, USA, 2003; Volume 304, 15p.
- 99. Aurich, J.C.; Mannweiler, C.; Schweitzer, E. How to design and offer services successfully. *CIRP J. Manuf. Sci. Technol.* 2010, 2, 136–143. [CrossRef]
- Sundin, E.; Ölundh Sandström, G.; Lindahl, M.; Öhrwall Rönnbäck, A. Using company–academia networks for improving product/service systems at large companies. In *Introduction to Product/Service-System Design*; Springer: London, UK, 2009; pp. 185–196.
- 101. Tukker, A.; Tischner, U. New business for old europe: Product-service development. Competitiveness and Sustainability 2006.
- Grieco Mazzine, E., Jr.; Benitez Nara, E.O.; Mahlmann Kipper, L.; Mairesse Siluk, J.C.; Rediske, G. Verifying practical design on furniture industry. *Int. J. Qual. Reliab. Manag.* 2015, 32, 881–894. [CrossRef]
- 103. Song, H.; Zhu, N.; Xue, R.; He, J.; Zhang, K.; Wang, J. Proof-of-Contribution consensus mechanism for blockchain and its application in intellectual property protection. *Inf. Process. Manag.* **2021**, *58*, 102507. [CrossRef]
- Akaba, T.I.; Norta, A.; Udokwu, C.; Draheim, D. A framework for the adoption of blockchain-based e-procurement systems in the public sector. In *Conference on E-Business, e-Services and e-Society*; Springer: Cham, Switzerland, 2020; pp. 3–14.
- Siregar, I.; Rizkya, I.; Syahputri, K.; Sari, R.M.; Anizar; Ariani, F.; Pintoro, A. Priority of selection suppliers with fuzzy ANP COPRAS-G. J. Phys. Conf. Ser. 2019, 1230, 12055. [CrossRef]
- Wang, S. Application of Product Life Cycle Management Method in Furniture Modular Design. *Math. Probl. Eng.* 2022, 7192152. [CrossRef]
- 107. Audy, J.; D'Amours, S.; Rousseau, L. Cost allocation in the establishment of a collaborative transportation agreement-an application in the furniture industry. *J. Oper. Res. Soc.* **2010**, *61*, 1559. [CrossRef]
- 108. Djunaidi, M.; Sholeh, M.A.A.; Mufiid, N.M. Analysis of Green Supply Chain Management Application in Indonesian Wood Furniture Industry; AIP Publishing LLC: Melville, NY, USA, 2018; p. 20050.
- Luo, B.; Sun, Y.; Shen, J.; Xia, L. How does green advertising skepticism on social media affect consumer intention to purchase green products? J. Consum. Behav. 2020, 19, 371–381. [CrossRef]
- 110. Castka, P.; Corbett, C.J. Governance of eco-labels: Expert opinion and media coverage. J. Bus. Ethics 2016, 135, 309–326. [CrossRef]
- 111. Ding, Y.; Luo, D.; Xiang, H.; Liu, W.; Wang, Y. Design and implementation of blockchain-based digital advertising media promotion system. *Peer Netw. Appl.* **2021**, *14*, 482–496. [CrossRef]
- 112. Namsawat, O.; Rugwongwan, Y. Tendency and Behaviour of Furniture Usage of Consumers in Thailand for Business Strategy Formation for Sustainable Environment. *Environ. -Behav. Proc. J.* **2017**, *2*, 27–34. [CrossRef]
- 113. Lin, C.R.; Chen, M.; Tseng, M.; Chiu, A.S.; Ali, M.H. Profit Maximization for Waste Furniture Recycled in Taiwan Using Cradle-to-Cradle Production Programming. *Math. Probl. Eng.* **2020**, 2020, 2948049. [CrossRef]
- Wong, W.P.; Deng, Q.; Tseng, M.; Lee, L.H.; Hooy, C.W. A stochastic setting to bank financial performance for refining efficiency estimates. *Intell. Syst. Account. Financ. Manag.* 2014, 21, 225–245. [CrossRef]
- Chang, F.; Zhou, G.; Zhang, C.; Ding, K.; Cheng, W.; Chang, F. A maintenance decision-making oriented collaborative cross-organization knowledge sharing blockchain network for complex multi-component systems. J. Clean. Prod. 2021, 282, 124541. [CrossRef]
- 116. Han, J.; Heshmati, A.; Rashidghalam, M. Circular economy business models with a focus on servitization. *Sustainability* **2020**, 12, 8799. [CrossRef]
- Ranta, V.; Aarikka-Stenroos, L.; Väisänen, J. Digital technologies catalyzing business model innovation for circular economy— Multiple case study. *Resour. Conserv. Recycl.* 2021, 164, 105155. [CrossRef]
- 118. Antikainen, M.; Uusitalo, T.; Kivikytö-Reponen, P. Digitalisation as an enabler of circular economy. *Procedia CIRP* **2018**, *73*, 45–49. [CrossRef]
- Quintais, J.P.; Groeneveld, L.; Bodó, B. *Blockchain Copyright Symposium: Summary Report*; The Institute for Information Law (IViR); Kluwer Copyright Blog: Amsterdam, The Netherlands, 2017; Available online: https://pure.uva.nl/ws/files/28023221 /Blockchain_Copyright_Symposium_Summary_Report_Kluwer_Copyright_Blog.pdf (accessed on 18 December 2022).
- 120. Kamble, S.; Gunasekaran, A.; Arha, H. Understanding the Blockchain technology adoption in supply chains-Indian context. *Int. J. Prod. Res.* **2019**, *57*, 2009–2033. [CrossRef]
- 121. Apte, S.; Petrovsky, N. Will blockchain technology revolutionize excipient supply chain management? *J. Excip. Food Chem.* **2016**, 7, 910.
- 122. Luisser, F.S.; Rosen, M.A. Improving the sustainability of office partition manufacturing: Balancing options for reducing emissions of volatile organic compounds. *Sustainability* **2009**, *1*, 234–253. [CrossRef]
- 123. Zheng, Z.; Xie, S.; Dai, H.; Chen, X.; Wang, H. An Overview of Blockchain Technology: Architecture, Consensus, and Future Trends. In Proceedings of the 2017 IEEE International Congress on Big Data (BigData Congress), Honolulu, HI, USA, 25–30 June 2017; pp. 557–564. [CrossRef]
- 124. Pilkington, M. Blockchain Technology: Principles and Applications; Edward Elgar Publishing: Cheltenham, UK, 2016.

- 125. Cortes Chavez, F.; Avila Chaurand, R.; Landa Avila, I.C. Effect of Subjective Evaluation Factors on the Buying Decision of Residential Furniture. *Procedia Manuf.* 2015, *3*, 6467–6474. [CrossRef]
- Li, Z.; Liu, X.; Wang, W.M.; Barenji, A.V.; Huang, G.Q. CKshare: Secured cloud-based knowledge-sharing blockchain for injection mold redesign. *Enterp. Inf. Syst.* 2019, 13, 1–33. [CrossRef]
- 127. Riesco, R.; Larriva-Novo, X.; Villagrá, V.A. Cybersecurity threat intelligence knowledge exchange based on blockchain. *Telecommun. Syst.* **2020**, *73*, 259–288. [CrossRef]
- 128. Patrickson, B. What do blockchain technologies imply for digital creative industries? Creat. Innov. Manag. 2021, 30, 585–595. [CrossRef]
- 129. Maesa, D.D.F.; Mori, P. Blockchain 3.0 applications survey. J. Parallel Distrib. Comput. 2020, 138, 99–114. [CrossRef]
- Wright, A.; De Filippi, P. Decentralized Blockchain Technology and the Rise of Lex Cryptographia. SSRN 2580664. 2015. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2580664 (accessed on 30 August 2021).
- 131. Zheng, Y.; Boh, W.F. Value drivers of blockchain technology: A case study of blockchain-enabled online community. *Telemat. Inform.* **2021**, *58*, 101563. [CrossRef]
- 132. Pazaitis, A.; De Filippi, P.; Kostakis, V. Blockchain and value systems in the sharing economy: The illustrative case of Backfeed. *Technol. Forecast. Soc. Chang.* 2017, 125, 105–115. [CrossRef]
- Dai, W.; Xiao, D.; Jin, H.; Xie, X. A Concurrent Optimization Consensus System Based on Blockchain. In Proceedings of the 2019 26th International Conference on Telecommunications (ICT), Hanoi, Vietnam, 8–10 April 2019; pp. 244–248.
- 134. Dounas, T.; Lombardi, D.; Jabi, W. Towards blockchains for architectural design consensus mechanisms for collaboration in BIM. In Proceedings of the 37th eCAADe and XXIII SIGraDi Joint Conference, Porto, Portugal, 9–13 December 2019.
- Posavec, A.B.; Aleksic-maslac, K.; Tominac, M. Non-fungible tokens: Might learning about them be necessary? In Proceedings of the 2022 45th Jubilee International Convention on Information, Communication and Electronic Technology (MIPRO), Opatija, Croatia, 23–27 May 2022; pp. 700–705.
- 136. Bamakan, S.M.H.; Nezhadsistani, N.; Bodaghi, O.; Qu, Q. Patents and intellectual property assets as non-fungible tokens; key technologies and challenges. *Sci. Rep.* 2022, *12*, 2178. [CrossRef]
- 137. Zhang, L. MRA: Metaverse Reference Architecture. Internet Things 2022, 12993, 102–120. [CrossRef]
- 138. Khan, F.; Kothari, R.; Patel, M.; Banoth, N. Enhancing Non-Fungible Tokens for the Evolution of Blockchain Technology. In Proceedings of the 2022 International Conference on Sustainable Computing and Data Communication Systems (ICSCDS), Erode, India, 7–9 April 2022; pp. 1148–1153.
- 139. Zhao, J.; Peng, Z. Shared Short-Term Rentals for Sustainable Tourism in the Social-Network Age: The Impact of Online Reviews on Users' Purchase Decisions. *Sustainability* **2019**, *11*, 4064. [CrossRef]
- 140. Joshi, Y.; Rahman, Z. Factors affecting green purchase behaviour and future research directions. *Int. Strateg. Manag. Rev.* 2015, *3*, 128–143. [CrossRef]
- 141. Black, I.R.; Cherrier, H. Anti-consumption as part of living a sustainable lifestyle: Daily practices, contextual motivations and subjective values. *J. Consum. Behav.* 2010, *9*, 437–453. [CrossRef]
- 142. Esmaeilian, B.; Sarkis, J.; Lewis, K.; Behdad, S. Blockchain for the future of sustainable supply chain management in Industry 4.0. *Resour. Conserv. Recycl.* 2020, 163, 105064. [CrossRef]
- 143. Planing, P. Business model innovation in a circular economy reasons for non-acceptance of circular business models. *Open J. Bus. Model Innov.* **2015**, *1*, 1–11.
- 144. Simões, P.; Marques, R.C. On the economic performance of the waste sector. A literature review. J. Environ. Manag. 2012, 106, 40–47. [CrossRef]
- 145. Bloom, P.N. How Will Consumer Education Affect Consumer Behavior? ACR N. Am. Adv. 1976, 3, 208–212.
- Cai, Z.; Xie, Y.; Aguilar, F.X. Eco-label credibility and retailer effects on green product purchasing intentions. *For. Policy Econ.* 2017, *80*, 200–208. [CrossRef]
- 147. Rahbar, E.; Wahid, N.A. Investigation of green marketing tools' effect on consumers' purchase behavior. *Bus. Strategy Ser.* 2011, 12, 73–83. [CrossRef]
- 148. Nielsen, A. Sustainable Efforts and Environmental Concerns around the World: A Nielsen Report; The Nielsen Company LLC: New York, NY, USA, 2011; Volume 11, p. 9.
- 149. Saberi, S.; Kouhizadeh, M.; Sarkis, J.; Shen, L. Blockchain technology and its relationships to sustainable supply chain management. *Int. J. Prod. Res.* 2019, *57*, 2117–2135. [CrossRef]
- 150. Andoni, M.; Robu, V.; Flynn, D.; Abram, S.; Geach, D.; Jenkins, D.; McCallum, P.; Peacock, A. Blockchain technology in the energy sector: A systematic review of challenges and opportunities. *Renew. Sustain. Energy Rev.* **2019**, *100*, 143–174. [CrossRef]
- 151. Kouhizadeh, M.; Zhu, Q.; Sarkis, J. Blockchain and the circular economy: Potential tensions and critical reflections from practice. *Prod. Plan. Control* **2020**, *31*, 950–966. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.